# National Representation and Local Public Expenditure: A Natural Experiment from Japan<sup>\*</sup>

Haishan Yuan<sup>†</sup>

University of Maryland, College Park Job Market Paper

January 27, 2014

#### Abstract

In Japan's mixed-member electoral system, a candidate who fails to obtain a pluarity of votes in a district may still be elected through a party list, giving her district two representatives instead of one. Extending the regression discontinuity design, I construct a sample of districts in which the assignment of an additional representative is as if random. I find that having an additional representative on average increases municipal expenditure by 1.8%. Within marginally winning districts, core municipalities of the second representative gain, but so do core municipalities of the first representative. This suggests that even in parliamentary systems with strong parties, political competition incentivizes politicians to bring public spending to core supporters in their districts.

Keywords: Political Representation, Public Expenditure, Mixed-Member Electoral System.JEL Classification Numbers: P16, H77, H72, D72.

<sup>\*</sup>The latest version can be found at http://econweb.umd.edu/~yuan/rep\_jp.pdf

<sup>&</sup>lt;sup>†</sup>Address: 3114 Tydings Hall, University of Maryland, College Park, MD 20742; Email: *yuan@econ.umd.edu*; I am grateful to Ethan Kaplan, John Shea, Allan Drazen, John Wallis, Razvan Vlaicu, Nuno Liñao, Jessica Goldberg, Lesley Turner, Olle Folke, Melissa Kearney, Peter Murrell, Judy Hellerstein, Ginger Jin, John Rust, Kei Kawai, Orhan Torul, Ruben Enikolopov, Nageeb Ali, Laurent Bouton, Yaniv Reingewertz, Lixing Tang, Ben Zou, Alex Whalley, Tuan Hwee Sng, Real Arai, Shigeo Hirano, Michio Umeda, Baolian Wang, Shulin Wee, Ron Chan and participants of the University of Maryland microeconomics brown bag seminar, 2013 Australasian Meeting of the Econometric Society, 2013 Asia Meeting of the Econometric Society, 2013 Singapore Economic Review Conference for their helpful comments. I thank Kei Kawai, Ben Zou and Yiwen Zhou for their help with obtaining data. All errors are mine.

## 1 Introduction

Does the number of political representatives affect the allocation of public resources? In a representative democracy, the primary means for citizens to affect policies is through their representatives. Therefore, the number of representatives is often considered a proxy of political power possessed by a group of citizens. For example, each state has two senators in the U.S. Senate, regardless of population. Small states therefore have greater representation in the Senate in percapita terms. Altas et al. (1995) and Lee (1998) find that per-capita federal spending is larger in smaller states. However, it is unclear whether the positive relationship between the number of representatives per capita and public spending is causal. And if so, what is the causal mechanism? In particular, since politicians are motivated or constrained by elections, what roles does electoral incentive have in translating greater political representation into more public spending?

In this paper, I extend the conventional regression discontinuity estimator applied to vote shares to estimate the effects of having an additional representative on local public expenditure in Japan. In Japan's mixed-member electoral system, a candidate who fails to obtain a plurality of votes in a single-winner district may still be elected through a party list, effectively giving her district two representatives instead of one. I find that having an additional representative on average increases total municipal expenditure by 1.8% and discretionary spending on public works by 7.7%. The higher expenditures are attributable to more transfers from the central government.

Moreover, within districts that are gaining representation, municipalities with a large share of supporters for the additional representative gain, but so do municipalities with strong support for the first representative. Because the second representative is likely to compete with the first representative in the following election, the presence of an extra representative weakens the incumbency advantage of the first representative, intensifying electoral competition. This result suggests that political competition incentivizes politicians to bring public spending to core supporters in their districts. I provide some evidence that strongholds for either the first representative or the additional representative have higher voter turnout rates in the following election. This is consistent with politicians delivering electorally motivated spending in order to turn out their core supporters to vote in the future elections.

The main results described above are obtained from a quasi-randomized sample I construct by

extending the conventional regression discontinuity design. I exploit two sources of discontinuity in Japan's electoral system. A candidate who loses in a district may still be elected if her ranking on the party list is good enough. Her ranking on the party list depends on her performance in the district race related to the winner there. A small change in the vote share of either the losing candidate or the winning candidate may alter the ranking of the losing candidate on the party list. This is the first source of discontinuity that I exploit.

The second discontinuity that I exploit comes from the ripple effects of the outcomes of close elections. A candidate who narrowly loses in a close election is likely to have a high ranking on her party list. If due to a small electoral shock the candidate instead wins in the close election, she vacates her position on the party list, allowing another candidate on the same party list to be elected. A close election in one district may therefore create a ripple effect on whether other districts would have an additional representative.

In a conventional regression discontinuity design, two candidates compete for office. Whichever candidate obtains more than 50% of the votes wins. The identification assumption is that in elections where candidates' vote shares are sufficiently close to the 50% threshold, the assignments of winners are as if random because a small random shock could alter them. I generalize this idea by perturbing the observed vote shares slightly to generate a counterfactual allocation of extra representatives to districts. Essentially, I construct a quasi-randomized sample, which consists of districts that may marginally gain or lose an additional representative when subjected to small perturbations to observed vote shares. In this sample, whether a district has an additional representative is as if random.

The natural experiment I analyze in this paper relates to three strands of literature on political representation, electoral competition and distribution of public spending. First, it relates to the literature that examines the empirical relationship between the number of representatives per capita and public spending. In Altas et al. (1995) and Lee (1998), since the number of seats for an electoral district is fixed, cross-sectional variation in the number of representatives per capita is driven by variation in population. But population potentially correlates with a large number of factors affecting public expenditure, creating difficulties for a causal inference. Ansolabehere et al. (2002) adopt a novel difference-in-difference (DID) strategy, using an arguably exogenous Supreme Court decision mandating the one-person-one-vote principle in the apportionment of state legislatures to

eliminate potentially confounding time-invariant heterogeneity. However, a mandated equalization of apportionment may change the power structure in the state legislature. For example, previously underrepresented urban interests may gain seats in the state legislature at the expense of rural interests. In this case, transfers to a local area may change even if its representation as measured by the seat-to-population ratio remains unchanged. In other words, the DID assumption of parallel trends may be violated.

Similar limitations also apply to the DID strategy adopted by Horiuchi and Saito (2003). They use changes in seat-to-population ratio due to the 1994 electoral reform in Japan. In this case, not only was the entire electoral system overhauled, but campaign finance regulations were also reformed to favor parties over individual politicians. For example, public subsidies to parties for campaigns and general administration were introduced, and corporate and labor union contributions to individual politicians were banned. Both the power structure of the legislature and the electoral incentives are likely to change after the electoral reform.

In this paper, I examine the effects of having an extra representative on public spending, holding the population and composition of an electoral district unchanged. The empirical strategy allows me to take power structure in the legislature as given, isolating the effects of effective representation on public spending in local areas. This paper adds to an emerging empirical literature that uses regression discontinuity designs to study political representation and policy outcomes. Albouy (2013) studies the effects of political representation in the majority party in the U.S. Congress on the federal grants received by states. Pattersson-Lidbom (2008) studies the effects of majority control by left-wing parties in Swedish local governments on taxation and government expenditure. Folke (forthcoming) studies the effects of partisan representation in Swedish local governments on local immigration, environmental and tax policies. This paper differs from Pattersson-Lidbom (2008) and Folke (forthcoming) by studying distributive policies by the national government, and differs from Albouy (2013) by focusing on the effects of number of representatives on local public expenditure in a parliamentary country.

Second, this paper relates to the literature on political agency problems. Incumbency advantage is often considered undesirable because it allows politicians to be less responsive to voters (Besley and Burgess, 2002). However, in a political agency model with both moral hazard problems and adverse selection problems, incumbency advantage naturally arises as voters are to some extent able to select better politicians through previous elections. The challenge is to empirically disentangle the incentive effects of incumbency advantage from the selection effects of incumbency advantage. In this paper, the quasi-randomized assignment of an additional representative to a district constitutes a negative shock to the incumbency advantage of the first representative. When a district is exogenously assigned with a second representative, such shock has no effect on the selection of the first representative. However, the disincentive effect of incumbency advantage for the first representative is weakened because the first representative is likely to compete with another incumbent in the following election. This result suggests that electoral competition is of first order importance in translating greater representation into more public spending.

Third, this paper adds to the literature on electoral rules and public finance. Electing legislators from small, single-winner districts holds politicians individually accountable, which may limit corruption (Persson et al., 2003) and help select better politicians (Besley, 2007). Legislators elected in large electoral districts that use party lists and proportional representation rules represent broad interests rather than narrow, geographically defined interests (Persson and Tabellini, 2000; Milesi-Ferretti et al., 2002). Combining two sets of electoral rules, a mixed-member system is often thought to have the best of both worlds and has been adopted in more than a dozen new and existing democracies (Shugart and Wattenberg, 2001). However, this paper documents a subtlety in the design of a mixed-member electoral system, which makes representatives elected through party lists responsive to narrow, geographically defined interests. In Japan, the supposedly broad representation of representatives elected from party lists is compromised by the linkage between candidates' ranking on the party lists and their performance in their small home districts.

The remainder of the paper is organized as follows. Section 2 provides institutional backgrounds and data description. Section 3 describes my empirical strategy. Section 4 presents the main estimates on the effects of political representation on local public finance. Section 5 discusses the role of political affiliation and legislative bargaining in driving the main results. Section 6 discusses how having an additional representative affects local public expenditure in more heterogeneous districts. Section 7 shows how having an additional representative differentially affects swing and core municipalities within a district. Section 8 concludes.

# 2 Institutional Background and Data

### 2.1 Public Finance in Japan

Figure 1 shows, for both Japan and the U.S., the shares of non-defense expenditures by the national government, locally financed expenditures by subnational governments, and expenditures by subnational governments financed by transfers from the national government. Compared to the U.S., subnational governments of Japan account for a relatively high share of total public expenditure. In Japan, 59% of non-defense public expenditures in Fiscal Year (FY) 2007 were spent by subnational governments, as compared to 52.2% in the U.S. Because in both Japan and the U.S., intergovernmental transfers amount to about 10% of total non-defense public expenditure, this means that the national government of Japan transfers a larger share of its revenue to subnational governments. However, subnational governments in Japan have very limited autonomy in generating local tax revenue (Weese, 2012). Prefectural and municipal governments rely on the national government as a major source of revenue.<sup>1</sup> From FY 2002 to FY 2010, total tax revenue of municipal governments on average accounted for 37.2% of total expenditure by municipal governments. Transfers from the national government and the prefectural governments accounted for 33.3% and 4.7%, respectively. Other sources such as debt, user fees and revenue from governmental enterprise made up the rest. Figure 2 shows these shares over the sample period. Because large municipalities are more capable of generating local tax revenue, the median municipality relies even more on transfers from the national government than those averages would suggest. The median share of municipal revenue due to transfers from the national government is between 40% to 50%from FY 2002 to FY 2010. Figure 3 shows the median shares of municipal revenue from various sources over this period.

Transfers from the national government are implemented by a tax sharing system. Several programs distribute funds to municipal governments, including the Local Allocation Tax, National Treasury Disbursements and the Local Transfer Tax. The Local Allocation Tax (LAT) is a formulabased general-purpose grant program that transfers fixed percentages of revenue of several major national taxes to municipal governments. To calculate the LAT transfer, national agencies take

<sup>&</sup>lt;sup>1</sup>As the immediately subnational administrative divisions, prefectural governments are analogue to state governments in the U.S. Prefectures are further divided into a number of municipalities.

the difference between the cost of providing a basic public services prescribed by law and the fiscal capacity of a municipal government. The National Treasury Disbursements provide mandatory cost sharing of certain public services, the cost of performing responsibilities of the national government entrusted to local governments and support of specific policies. Finally, the Local Transfer Taxes transfer a fixed proportion of revenue of several national taxes, mostly excise taxes, to local governments. Though these programs are more or less formula-based, numerous factors and discretionary adjustments are considered. Anecdotal evidence suggests that the bureaucratic application of transfer formulas is not carried out entirely free of political interventions. For example, politicians in the late the 1990s and early 2000s successfully lobbied the Ministry of Home Affairs to include access to high-speed rail as a basic public service, allowing the use of LAT grants to fund bullet train expansion projects in remote areas (DeWit, 2002).

Given the municipal governments' fiscal reliance on the national government, it is reasonable to use total municipal expenditures to measure the effect of political representation in the national government on local public finance. However, I also study the effect of political representation on transfers from the national governments, the gap between total public expenditure and local tax revenue, and more disaggregate spending measures, although data for these measures are more limited.

#### 2.2 Japan's Mixed-Member Electoral System

From 1947 to 1993, Japan had an electoral system featuring multi-member districts (MMD) and a single, non-transferable vote (SNTV). The nation was divided into more than a hundred mediansize districts. Each district elected two to six members to the House of Representatives, the lower house of Japan's parliament, the National Diet, for a term of four years. Candidates in each district with the highest vote count would be elected. The upper house, the House of Councillors, was elected through a similar multi-member district plurality rule, though its members were elected from larger districts for a longer term. Typically, then and now, a majority coalition in the House of Representatives forms the government and elects one of its members as prime minister. The prime minister can dissolve the House of Representatives before its term expires and call for early elections, but not for the House of Councillors. While the House of Councillors retains considerable legislative power, the House of Representatives prevails in disagreements between the two chambers on important decisions such as passing a budget, ratifying treaties and choosing a candidate for prime minister. Moreover, the lower house can override the upper house's objection on a regular bill by a two-thirds majority. Given the dominant role of the House of Representatives, Japan's constitutional design is referred to as a "one-and-a-half house solution" by Ackerman (2000). I shall focus on the House of Representatives in this paper.

The Liberal Democratic Party (LDP) had been the ruling party since 1955. However, in the 1993 general election, it lost its governing position for the first time. A governing coalition was formed by eight small anti-LDP parties. Led by Prime Minister Morihiro Hosokawa in 1994, the 11-month governing coalition replaced the previous MMD-SNTV system with a mixed system for the lower house.

Under the reform, the House of Representatives was given 500 seats, of which 300 seats were from single-member districts (SMDs) with a first-past-the-post (FPTP) rule, while 200 seats were elected from proportional representation (PR) party lists grouped by 11 regional PR blocks.<sup>2</sup> Under the new system, each voter is given two votes, one for a candidate in her single member district and another for a party list in her PR block. The SMD vote need not be for a candidate from the same party as the PR vote. The boundaries of PR blocks do not cross the boundaries of prefectures. the immediate sub-national level of administrative unit. Conversely, the boundaries of prefectures do not cross the boundaries of SMDs. Hence, a PR block contains one or several prefectures, and a prefecture contains several SMDs. Figure 4 is a map showing how Japan is divided into 11 PR blocks, each filled with different colors. Each PR block consists of one or several prefectures, as delineated by dashed lines. The number of SMDs in each prefecture in the most recent 2012 election is labeled on top. A fixed number of PR seats is allocated to a PR block before each election. Parties propose a party list in each PR block to contest for the PR seats allocated to that block. PR seats in a PR block are allocated to parties in proportion to their PR vote shares in the block. Vote shares outside a PR block have no bearing on the allocation of the PR seats within the PR block.

As in some other mixed-member systems, such as those of Germany and New Zealand, dual candidacy is permitted. A candidate can be on both the SMD ballot and the PR list ballot. If

 $<sup>^{2}</sup>$ After the 1996 election, the first after the reform, the number of PR seats was reduced to 180, while the number of SMD seats remained unchanged.

a candidate wins a seat from an SMD, she takes that seat and vacates her position on the party list. If a candidate loses in the SMD race, she can still obtain a PR seat if her ranking on the PR list is favorable relative to the number of PR seats her party won in the regional PR block. In Germany and New Zealand, the allocations of PR seats are used to top up district seats so that the overall shares of seats going to each party at the end would proportionally reflect vote shares of parties nationwide. But in Japan's system, the number of SMD seats (or constituent seats) and the number of PR seats are fixed. The SMD system and the PR system are parallel in the sense that the number of PR seats a party obtains only depends on its performance in the PR vote and the number of SMD seats a party obtains only depends on its candidates' performance in SMD races.

To illustrate, compare Germany and Japan. The Bundestag, the German Federal Diet, consists of 598 members, with half elected from single member districts and the other half from party lists proportionally allocated to parties according to nationwide party vote shares. If a party obtains 50% of party votes nationwide and its nominees win 100 seats in the single member districts under the first-past-the-post rule, the number of PR seats allocated to the party is  $199 = 598 / 2 - 100.^3$  In Japan, on the other hand, the number of PR seats a party wins is the sum of PR seats won in each PR block, which in turn is determined solely by PR vote shares in each block, independent of the number of SMD seats won or nationwide PR vote shares.<sup>4</sup> The number of PR seats, prefectures and SMDs in each PR block are summarized in Table 1.

While the SMD system and PR system are parallel in that the allocation of PR seats across parties does not depend on the outcomes of SMD races, the two systems are connected in the allocation of intra-party PR seats. A special feature of the party PR lists is that rankings are partially determined ex ante and partially determined ex post. Candidates on the PR lists are ranked by their parties before the election. However, parties can give multiple candidates equal ranks on the ballot. Dual candidates' ex post ranks within a cluster (conditional on ex ante equal rank) are determined by their performance in their own SMD, specifically by their vote share divided by the winning candidate's SMD vote share. The higher is this ratio (hereafter known as

 $<sup>^{3}</sup>$ In the case that a party has more members elected from SMDs than its overall seat share implied by the national party vote share, some additional seats known as overhang seats are added to the 598 regular seats to accommodate the crowd-out of PR seats for other parties that would otherwise occur.

<sup>&</sup>lt;sup>4</sup>The mapping from PR vote share to PR seat share in a PR block follows the D'Hondt method.

the narrowness-of-defeat ratio or simply the narrowness ratio), the higher is a dual candidate's rank within the cluster. For example, suppose all candidates on a party list are dual candidates who lose their SMD races. Amy ranks first on the list, which is fixed ex ante, but Ben, Cameron and David rank equally second. The ex-post ranks of Ben, Cameron and David will be given by their vote shares in the SMD races as compared to the winning candidates from the respective SMDs. If the party of Amy and Ben obtains two seats in the PR block, Amy will get a PR seat regardless of her narrowness ratio, while the candidate among Ben, Cameron and David with the highest narrowness ratio will get the second seat.<sup>5</sup>

If legislators who are defeated in the SMD races but elected through the PR system are motivated to maintain a local base, such a parallel voting system creates variation of *de facto* representation in the lower house across districts. There are several reasons candidates would have incentives to cater to local interests (McKean and Scheiner, 2000). First, dual candidacy provides insurance to candidates for a seat via an alternative route. If a candidate loses the SMD race, she may still be able obtain a seat through the party list. Secondly, the PR list is partially open in that a dual candidate's ranking on the list is partially determined by her performance in the SMD. Third, new formation of parties and changes of party membership are relatively frequent in Japan. The Democratic Party of Japan, which recently lost its majority in the 2012 general election, was founded in 1998. The current third largest party, the Japan Restoration Party with 54 seats, and the sixth largest party, the Tomorrow Party of Japan with nine seats, were both founded in 2012. A local base provides politicians with political capital and puts them in a good bargaining position should change of party affiliations occur. Fourth, it is not uncommon in Japan's political culture that seats are inherited by staffers or children of the incumbents (Taniguchi, 2008). A local base would facilitate such inheritance.

#### 2.3 Data

There have been six general elections for the House of Representatives since the electoral reform in 1994. They were held in years 1996, 2000, 2003, 2005, 2009 and 2012. I downloaded election

<sup>&</sup>lt;sup>5</sup>There is one caveat. After the 1996 general election, the election law was amended such that any candidate who fails to obtain a 10% vote share in the SMD race will be disqualified. Her position on the PR list would be vacated regardless of her narrowness ratio, and her deposition for candidacy would be forfeited. This has been taken into consideration in the implementation of my empirical strategy.

results and party lists from the website go2senkyo.com for all six elections and the website of the Japanese Ministry of Internal Affairs and Communications for the last three. I obtained municipal level voting data from 1996 to 2005 from Asahi Shimbun, one of five major national newspapers in Japan. Municipal election data for 2009 were complied in part from various websites of prefectural election commissions. I mainly used the election data from the website go2senkyo.com, as they were more complete, but verified them with data from the other two sources, finding few discrepancies.

In these elections, among candidates who lost their SMD races but ran again in the next election. 88% of challengers who were not members of the Lower House ran again in the same district, while PR incumbents were 8.6% more likely than non-members to run in the same SMD. This difference is statistically significant, as shown in Table 2 using a linear probability model. A logit model gives a similar result. This suggests that most candidates run in the same district if they run again in the next election, and that PR incumbents are particularly inclined to do so.<sup>6</sup> Fifty-nine percent of candidates contesting for SMD seats have been dual candidates. The percentage has been higher among competitive candidates; 84% of candidates finishing first or second in SMD races have been dual candidates. Moreover, 81% of candidates on the party lists have also been on the ballot of an SMD race, suggesting a preference for being elected to an SMD seat. This is consistent with SMD incumbents having a higher re-election rate than do PR incumbents. While incumbents of PR seats enjoy substantial electoral advantages, incumbents of SMD seats have considerably higher probability of getting re-elected to the lower house. Based on a linear probability model relating re-election probability to incumbent status, controlling for party-election fixed effects, the incumbency advantage of SMD incumbents is about 63% to 157% higher than that of incumbents elected through the party lists. These regression results are shown in columns (1) to (4) of Table 3. Even among incumbents who were elected in close SMD elections, so that the assignment to SMD seats versus PR seats is close to random, SMD incumbents had a higher rate of getting re-elected. This is shown in columns (5) and (6) of Table 3.

Demographic data and basic public finance data were taken from two sources: the Minryoku database and the Ministry of Internal Affairs and Communications. The Minryoku database was compiled by Asahi Shimbun Publications Inc. from various governmental agencies. Municipal public expenditure data and local tax revenue from FY 1997 to FY 2009 were available in the Minryoku

<sup>&</sup>lt;sup>6</sup>Unsurprisingly, incumbents of SMD seats almost always ran in the same district from which they were elected.

database. More detailed breakdowns of revenue and expenditure of municipal governments from FY 2002 to FY 2010 are available from the Ministry of Internal Affairs and Communications.

# 3 Empirical Strategy

#### 3.1 Sources of Discontinuities to be Exploited

In light of the above-described electoral rules linking the Majoritarian and the PR system, which are imposed uniformly across parties, there are two sources of discontinuity to be exploited for exogenous variation. The first is close narrowness ratios among ex ante equally ranked dual candidates. To illustrate this, consider the party list of LDP in the general election of 2009 for the PR Block of Kitakanto in Table 4.<sup>7</sup> Pure PR candidate Genichiro Sata occupied the singleton top rank on the list. After Genichiro Sata, 26 candidates were ranked equally second. They were each dual candidates, competing in one SMD within the Kitakanto PR block. On the bottom of the list, two pure PR candidates were ranked 28th and 29th, respectively. Twenty seats were allocated to the PR Block of Kitakanto in 2009. LDP won 25.84% of party votes in this PR block, hence obtaining six seats according to the D'Hondt method. Genichiro Sata took up one PR seat by being on top of the list as a pure PR candidate, leaving five seats for candidates below him. Three dual candidates won in their respective SMDs, hence taking the SMD seats and vacating their positions on the party list. The SMD-losing dual candidates in the second-rank cluster were then ranked according to their narrowness-of-defeat ratio, i.e., their vote share divided by the vote share of the winner in their own district. The five candidates with highest narrowness ratios obtained the remaining PR seats. The narrowness ratio of Tsutomu Sato, who took the last PR seat for LDP in Kitakanto, was 0.781. Yuya Niwa, who had a narrowness ratio of 0.772 and was ranked ex post immediately below Tsutomu Sato, did not get a PR seat. In this case, Tochigi 4, the district of Tsutomu Sato, obtained an additional *de facto* representative through the PR system, while Ibaraki 6, in which Yuva Niwa competed, did not. Notice that, given the number of seats a party obtains, the cut-off narrowness ratio for the party's last PR seat is potentially determined by order statistics of the narrowness ratios from all equally ranked candidates in a PR list cluster, rather than a single number as in the FPTP two-party elections. Determination of the cut off depends on multiple vote

<sup>&</sup>lt;sup>7</sup>The PR Block of Kitakanto is north of Tokyo Prefecture, and is painted in orange in Figure 4.

counts among candidates whose identities are ex ante uncertain. Therefore, it would be difficult to engage in electoral manipulations just around the cut-off in order to gain the last PR seat and award an SMD an extra *de facto* representative. Endogenous sorting in a small neighborhood of the cut-off is highly unlikely, avoiding the most dangerous pitfall invalidating a traditional regression discontinuity design, particularly in studies examining two candidates contesting under the plurality rule. In the above example, if due to random factors Yuya Niwa had achieved an additional 0.5 percentage point in vote share, he would have obtained the last PR seat at the expense of Tsutomu Sato. Notice that Yuya Niwa actually had a higher vote share than Tsutomu Sato did. The reason he was not able to obtain the last PR seat is that the winner in Yuya Niwa's SMD did better than the winner of Tsutomu Sato's SMD. Had the winner of Yuya Niwa's SMD attained a  $\frac{3}{4}$  percentage point lower vote share, or had the winner of Tsutomu Sato's SMD achieved a  $\frac{3}{4}$  percentage point higher vote share, Yuya Niwa would have obtained the last PR seat instead of Tsutomu Sato. Thus, the winner of the last PR seat depended on at least four vote counts: Niwa's and Sato's votes and the votes of the winners in their districts. Moreover, the identities of these four vote counts are only relevant conditional on the LDP obtaining six PR seats and having four dual candidates with narrowness ratios higher than Tsutomu Sato did, both of which were uncertain before the election results were revealed.

The second source of discontinuity is close elections in SMD races. A narrow winner in one SMD could potentially change the representation of another district in the same PR block, because of its implications for the intra-party allocation of PR seats. This is because winners of SMD races vacate their positions on the party lists. To see this operating in reality, consider again the LDP's party list in Table 4. The SMD-losing candidate with highest narrowness ratio at 0.976 was Fukushiro Nukaga from district Ibaraki 2. If for random reasons he had obtained an additional 1.2 percentage points in vote share, he would have won the SMD seat and vacated his position on the PR list. The last PR seat would have then gone to Yuya Niwa. In such a scenario, the opponent of Fukushiro Nukaga, who would have lost the SMD race by a narrow margin, would have occupied a high ex post ranking on his party's PR list, potentially kicking out another SMD-losing candidate from that party. The outcome of a narrow election in SMD Ibaraki 2, though perhaps not consequential for its own representation, thus has a ripple effect on the representation of two other districts.

There is another source of discontinuity that can potentially be utilized. Folke (2011) proposes

a method of applying regression discontinuity design in proportional representation systems, exploiting the discontinuous jumps in the mapping of practically continuous vote shares to discrete seat shares. He then applies this method using Swedish municipal elections, which have a pure PR system, to study the effects of party representation on environmental, immigration and tax policies. The benefit of exploiting such discontinuity in Japan's case is that it would provide an extra source of exogenous variation of effective representation due to the marginal change of PR seats obtained by a party, which may lead to the election (or non-election) of SMD-losing dual candidates from the affected parties. One cost, however, is that this strategy would introduce another layer of complexity, as such discontinuity rests on the particulars of the mapping from the PR vote shares to the number of PR seats obtained by each party. More importantly, inter-party re-allocations of PR seats may have wider political and public policy implications than intra-party reallocations of PR seats across districts. As shown in Folke (2011), the assignment of a seat in the municipal legislature to parties with different agendas has large effects on local immigration and environmental policies. Inter-party re-allocations of PR seats may also alter the coalition formation, regional bargaining positions, public policy priorities and so on. Exploiting this discontinuity therefore confounds the distributional consequences of different levels of effective representation, holding the partian configuration of a legislature fixed. I shall focus instead on the cross-municipality variation of effective representation induced by intra-party assignment of PR seats.

## 3.2 A Quasi-Randomized Sample of Districts

To motivate how I incorporate the two sources of discontinuity in my empirical work, consider the following thought experiments. Imagine that due to random factors, such as weather on the election day affecting turnout of voters for candidates differentially, the vote shares of the winner and the runner-up candidate in a particular SMD are perturbed. In particular, suppose I transfer an amount  $\epsilon$  of vote share from one candidate to the other. This may or may not affect the outcome of the perturbed SMD race. If under this counter-factual vote share profile, the allocation of representatives to districts via the PR system does not change, nothing happens. However, if a district having exactly one additional representative through the PR system loses it in the counter-factual, the district is tagged as randomly assigned to the treatment of having two effective representatives. If a district having no additional *de facto* representative through the PR system gains one in the counter-factual, the district is tagged as randomly assigned to the control group of having a single representative. The counter-factual vote shares may result in changes in district representation due to either or both of the above-mentioned sources of discontinuity. To construct a sample of districts with quasi-randomly assigned treatment status, I carry out the concrete version of the above thought experiments on each SMD in each election, perturbing one SMD election at a time and holding everything else constant. This generates a set of treatment districts (i.e., districts having exactly one PR representative who would lose it in the counter-factual) and a set of control districts (i.e., districts having no PR representative who would gain one in the counter-factual). A district may qualify for treatment status under multiple perturbations to different elections, but such districts are not double counted in the quasi-randomization sample.

I focus on the margin of having zero or one PR representative, so that districts in the quasirandomized sample have identical and exactly one treatment status out of two, regardless of which thought experiment generates the treatment status. This avoids complexity arising from situations such as a district having a treatment status at the 0–1 margin but a control status at the 1–2 margin. Moreover, it is so rare that a district could gain or potentially gain two PR representatives that precise estimation at margins other than zero-one is difficult. Furthermore, it should be noted that a given perturbation does not always generate treatment and control districts in pair. It is possible that a vote share perturbation generates a treated district but not a control district, and vice versa, because a dual candidate may gain or lose a seat to a pure PR candidate.

In the regression discontinuity (RD) design of Lee et al. (2004), who study U.S. House elections, the authors suggest a non-parametric estimate using close elections with a margin of victory of less than 4% in the two-party vote share. In those elections, election outcomes are considered to be as if random. Since vote share transfers of up to 2% between the two candidates are sufficient to alter the outcomes in these elections, I similarly use 2% perturbations of vote shares to construct the quasi-randomized sample, i.e.  $\epsilon = 0.02$ . Table 5 shows how many SMDs have additional representatives after each election, in both the full sample and the quasi-randomized sample.

To check whether the constructed sample has close to random assignments of treatment status (i.e., having a PR representative or not), I examine the correlations between treatment status and a list of demographic and political variables. This list includes municipal population growth rate, area of the municipality, population density, number of SMD candidates, total vote share of the top two SMD candidates, whether the SMD elected a LDP candidate and whether the SMD elected a candidate of the Democratic Party of Japan (DPJ), which was the main opposition party for most of the sample period. Results are reported in Table 6. None but the dummy variable indicating a DPJ winner are found to be significantly correlated with the treatment status of a municipality at a 10% level.

# 4 Additional Representation and Local Public Finance

#### 4.1 Municipal Public Expenditure

To estimate the effect of having an additional *de facto* representative through the PR system on public expenditure, my main specification is

$$log(y_{it}) = \alpha + \delta P R_{it} + X'_{it}\beta + \mu_i + \pi_t + \epsilon_{it}$$
(1)

where  $y_{it}$  is the public expenditure per capita for municipality *i* in fiscal year *t*;  $PR_{it}$  is a dummy variable equal to one if municipality *i* has one or more SMD-losing but PR-elected representative at time *t* and zero otherwise;  $X_{it}$  is a vector of demographic and economic controls;  $\mu_i$  is a municipal fixed effect and  $\pi_t$  is a year fixed effect. First, in Table 7, I present coefficient estimates using the full sample of municipalities, except a few large municipalities that span multiple districts, from FY 1997 to FY 2010. It should be noted that municipalities are rarely split into multiple SMDs except when the municipality is very large in population. If time-invariant heterogeneity across municipalities is correlated with having an additional PR representative, but time-varying factors are not, this specification provides consistent estimates through the inclusion of municipal fixed effects. In this sample, municipalities with at least one PR representative have a public expenditure per capita 0.86% higher than comparable municipalities without a PR representative; this result is significant at the 5% level.<sup>8</sup> To control for economies of scale in public goods provision, the cost of providing public goods and the demand for public goods, I include log municipal population, log taxable

<sup>&</sup>lt;sup>8</sup>For comparison with the later results from the quasi-randomized sample, here  $PR_{it}$  is a dummy variable equal to one if municipality *i* has one or two SMD-losing but PR-elected representatives and zero otherwise. Adding another dummy variable indicating having two extra PR representative does not change the results. As reported in Table A.1, having two extra PR representatives rather than one further increases public expenditure, but its effect is imprecisely estimated due to the small number of districts having two PR representatives.

income per capita and population shares of age groups 0-4, 5-19 and 65+ as control variables. To control for and compare the traditionally estimated effect of mal-apportionment, I include the log voting population of the SMD the municipality belongs to. The estimates of the main representation effect are robust to the inclusion of these controls. Here, as in later estimations, standard errors are robust to two-way clustering on municipality and on PR block–House term. This allows for time series correlation within municipalities and cross-municipality correlation within a PR-block in a House term. There may be cross-municipality correlation within a PR block–House term because some municipalities share the same representatives or the elections of PR representatives are correlated within a PR block.<sup>9</sup>

One potential concern about these estimates is that there is some unobserved time-varying factor that correlates with both the public expenditure in a district and the probability that the district has a PR representative. For example, recognizing that a PR representative could bring in more public spending, a district with temporarily high demand for public spending may vote strategically for the runner-up to increase its chance of having a PR representative. If the high demand for public spending, say due to a natural disaster, would be partially fulfilled even in the absence of a PR representative, the estimated effect of having a PR representative on public expenditure would be biased upward. On the other hand, if a district's SMD representative is very successful at bringing in pork barrel spending and is rewarded electorally by voters, the narrowness ratio of the runner-up would be low and the district may not have a PR representative. If there is persistence in how much pork barrel spending an SMD representative brings, having a PR representative would be negatively correlated with the persistent unobserved component of municipal spending. Using the full sample, the estimated effect of having a PR representative on expenditure would be biased downward. To address such concerns, I estimate the effect of having a PR representative on public expenditure using the quasi-randomized sample described in the previous section. Notice that, even if voters are strategic as described above, this sample still provides a consistent estimate as long as voters are not able to coordinate precisely to gain a PR representative by foreseeing small electoral shocks. There are reasons to believe that voters are not that sophisticated. A district typically has more than 300 thousand eligible voters. It is extremely difficult for voters to coordinate precisely

<sup>&</sup>lt;sup>9</sup>One-way clustering, either on municipality or PR block–House term, results in smaller standard errors for most estimates reported in this paper.

to ensure that their SMD-losing candidate is sorted into one side of the cut-off for the PR seat in the face of small electoral shocks. Polling and forecasting prior to elections are not very extensive in Japan. For example, the Democratic Party of Japan was surprised by its own success in the 2009 general election. Had it listed two more names on its party list in the Kinki PR block, it could have obtained two more PR seats.

Compared with the full sample, SMD races in the quasi-randomized sample are more competitive. The average margin of victory is 11.5%, compared to 15.5% in the full sample. However, it should be noted that the most competitive districts are unlikely to be included in the quasirandomized sample. Eighty-four percent of runners-up are dual candidates, and among these, one third are elected to a PR seat. If two dual candidates, the winner and the runner-up, have roughly equal votes, whoever loses in the SMD race would have a high narrowness ratio and hence would rank high among her ex ante equally ranked peers. Thus, districts that are highly competitive will have a PR representative with a probability close to one regardless of who wins the SMD seat. A small perturbation of vote share would not be sufficient to deprive them of a PR representative. Similarly, very safe districts are excluded from the quasi-randomized sample because they require huge electoral shocks in order to elect one of their candidates through the PR system. Therefore, the quasi-randomized sample contains SMDs with meaningful but not most intensive electoral competition. Inferences based on this sample should be useful for addressing broader issues. For example, the exogenous variation of electoral strength studied in Lee et al. (2004) comes from the closest elections in the past and the incumbency advantages the narrow winners enjoy subsequently. They infer that electoral strength has limited influence on the voting records of legislators compared with a legislator's identity. In this context, voters rather vote to choose represented policies than to pressure representatives to chose their preferred policies. However, it need not apply to districts where elections are less competitive. If polarized districts have more competitive elections, policy moderation there may provoke backlash among an incumbent's base voters, harming their electoral prospects via primary challenges or lower turnout from core supporters. This would be consistent with Gerber and Lewis' (2004) finding that legislators' positions diverge more from the preference of median voters in more heterogeneous districts.

Table 8 reports estimates using the specification in equation (1), but with the quasirandomized sample. In the baseline specification with fixed effects but no other controls, municipalities with a PR representative are estimated to have on average 1.82% more public expenditure per capita, which is significant at the 1% level. The magnitude and statistical significance of the main coefficient estimate remain stable with the addition of controls. Notice that in the quasi-randomized sample, controls are not in principle necessary for identification even if they are correlated with public expenditure; although in practice, adding controls might help to reduce noise and can help us assess the robustness of the estimates in the finite sample. In column (6) of Table 8, the voting population of the SMD containing the municipality has a coefficient of -0.2000, significant at the 5% level. It contrasts with the insignificant estimate of -0.0068 from the full sample. The instability of estimates for this coefficient may suggest that the size of the electorate or district population may proxies for other variables, or it may suggest that the size of electorate has a different impacts in different samples.

#### 4.2 Transfers from Central Government

Ideally, one would like to use data on discretionary transfers from the national government to the municipal governments as a dependent variable to confirm the political cause of higher municipal public expenditure. However, I only have categorical expenditure data for a subset of the sample period, and even in this data, discretionary transfers cannot be clearly identified. Moreover, targeted transfers may not be carried out transparently through discretionary items, but rather in a more disguised fashion by tampering with parameters used to determine transfers in various programs. To see if having an additional PR representative affects the total amount of transfers from the national government, I re-estimate Eq. (1) but instead with log per-capita transfers from the national government as the dependent variable. Using the quasi-randomized sample from FY 2002 to FY 2010, I find that having an additional PR representative has a positive effect of 1.7% to 1.9%, depending on specifications. However, these coefficients are imprecisely estimated. Standard errors robust to two-way clustering are about 1.1%, giving most estimates a *p*-value around 10% if the coefficient is tested against zero. Table 9 reports these results, as well as estimates using the full sample over the period FY 2002 – FY 2010. Results from the full sample are broadly similar, with estimates ranging from 0.87% to 0.93% and marginally significant.

Note that the estimates in Table 9 use only a subset of the time periods available in the original sample, due to data limitations. However, since transfers from prefectural governments

only account for a small share of revenue for municipal governments (see Figure 2), I can use the difference between total local public expenditure and local tax revenue to proxy the amount of transfers from the national government for a longer sample period. This alternative measure is highly correlated ( $\rho = 0.951$ ) with the direct measure of transfers over the period FY 2002 to FY2010, for which I have direct data on transfers from the national government. Estimation results using the alternative measure of transfers over the longer sample period FY 1997 to FY 2012 are reported in Table 10. In the full sample, estimates from various specifications suggest that an extra representative results in a 1.1% to 1.2% increase of transfers, with all estimates significant at the 5% level. In the quasi-randomized sample, the effect is much larger, ranging from 2.1% to 2.6%, again all significant at the 5% level. Therefore, these results provide evidence that the higher public expenditure associated with having a PR representative is due to more transfers from the central government.

#### 4.3 Robustness

To see whether the results are driven by small municipalities, I re-estimate the baseline results in Table 7 and Table 8 with each municipality weighted by their population. Results are shown in Table A.2. Estimates remain significant at conventional levels. In the full sample, the magnitude is slightly higher. With the quasi-randomized sample, the estimated magnitude drops modestly. One possible explanation is that smaller municipalities are easier to target for electorally motivated transfers. From 2003 to 2004, there was a large wave of municipal mergers (see, for example, Weese, 2011). The number of municipalities decreased from more than 3,200 to less than 2,000. While this should not affect the consistency of estimates from the quasi-randomized sample, and while the municipal fixed effects reflect any change of municipal identity, mergers may bias the estimates from the full sample if they are correlated with having a PR representative. For example, there is a trade-off in the number of jurisdictions between economies of scale in public goods provision and heterogeneity of preferences (Alesina and La Farrara, 2000; Weese, 2012). If economic integration after mergers increases public expenditure over time and if political integration increases a district's chance of having a PR representative, the estimated effect of having a PR representative may be biased upward. In estimates reported in the upper panel of Table A.3, I re-estimate the effect of having a PR representative on per capita public expenditure as well as the per capita gap between public expenditure and local tax revenue using a balanced panel of municipalities that are in the sample for the entire period. In this sample, no municipality is involved in any merger over the sample period. The effects are less precisely estimated, but are quantitatively similar.

In the quasi-randomized sample, a municipality enters the sample following an election in which it received a treatment status or control status for some perturbation and exits if it does not receive a treatment or control status in the next election. One may be concerned that municipalities with infrequent presence in the quasi-randomized sample are considerably different from other municipalities in the randomized sample and that such unobserved characteristics drive the results. For example, suppose that voters are aware that additional representation through the PR system is able to bring in additional funding from the national government, and vote strategically to lower the margin of victory for the SMD winner. This by itself would not invalidate my identification strategy as long as voters are not able to coordinate and precisely control the allocation of vote shares to their candidates. But if municipalities only exercise strategic voting when there is a high demand of public expenditure and such demand is met when a candidate who lost in that SMD is elected through the PR system, the estimated average treatment effect could be largely driven by these municipalities with a high treatment effect. I therefore re-estimate the treatment effect of having a PR representative using a sub-sample of the quasi-randomized sample, including only municipalities present in the sample for at least half of the sample period (i.e. seven years). The estimates shown in the second part of Table A.3 are quantitatively similar to the results in the full sample and remain significant at the 5% level.

In the full sample, local political characteristics may correlate with having a PR representative, hence confounding the estimated causal impact of representation. In the quasi-randomized sample, this is not a concern in principle as long as the vote share perturbations are small enough. To see whether political characteristics are a concern in a finite sample, I include an alternative set of political controls. They include the vote share of the SMD winner, the vote margin difference between the SMD winner and SMD runner-up and the narrowness ratio of the runner-up in the SMD. Results reported in Table A.4 show that the main estimates of interest are robust to the inclusion of these controls in either sample.

#### 4.4 Public Works, Welfare Expenditure and Government Payroll

While having an additional *de facto* representative increases total municipal public expenditure by about 1.8%, it is unclear the extent to which this represents an increase of discretionary spending. The Ministry of Finance classifies spending into three types: discretionary spending, compulsory spending and others. Discretionary expenditures are mainly on public works of infrastructure. Compulsory spending consists of debt service, wages and salaries of government employees, and welfare spending. Nation-wide, public works expenditures account for about 15%of total municipal expenditure. Welfare spending and government payroll account for 15% and 19%, respectively. Table 11 shows the estimated impact of having an additional representative on per-capita municipal expenditure on public works, welfare and payroll, respectively. Estimates from the quasi-randomized sample suggest that an additional representative increases public works spending by as much as 8%, while reducing welfare spending by about 2%. This is consistent with the view that public works in Japan often function as job support programs (Schlesinger, 1999). Municipal governments in Japan have little authority in setting welfare policies. The scope, eligibility criteria and payment standards for welfare are set by the national government in a fairly uniform manner, though with some regional adjustments reflecting variations in the cost of living. Because public works spending provide jobs and economic stimulus in local areas, fewer people would need or be eligible for welfare, which in turn lowers welfare expenditure administrated by the municipal government. Having a PR representative has no significant effect on the payroll of municipal employees. In unreported results, I do not find any significant effect on the numbers of temporary or permanent government employees either. This result suggests that higher expenditure is not driven by patronage spending through government employment. This result is consistent with the fact that, except in the largest cities like Tokyo or Osaka, most candidates for local governments are non-partial or are affiliated with the Japanese Communist Party, which has little presence in national politics. Therefore, local officials are unlikely to engage in partia politics by manipulating municipal employments.

# 5 Partisan Affiliation and Legislative Bargaining

In the legislative bargaining framework of Baron and Ferejohn (1989), representatives join a minimum winning coalition to gain rents. An agenda setter who proposes how to split a fixed pie is able to extract more rents. Members of the majority party often have a better chance to be recognized as agenda setters and, therefore, receive more rents. Albouy (2013) finds evidence supporting this model. He finds that states with a larger portion of their delegates belonging to the majority party of the United States Congress receive a larger amount of federal grants.

In the presence of party discipline, however, it is less clear whether a typical representative is able to receive a larger amount of rents if he or she is affiliated with the majority party or is a member of the governing coalition. The impact of party affiliation on rents depends on the source of party discipline. Since parties are both legislative institutions and electoral institutions, there are two interconnected but conceptually distinct aspects of party discipline, namely legislative party discipline and electoral party discipline (Myerson, 1997). With legislative party discipline, legislators are expected to vote with their party. With electoral party discipline, legislators rely on their party to be elected. For example, in a proportional representation system where voters vote only for parties and parties have full control over the ranking of candidates on a party list, electoral party discipline is extremely strong, as parties decide the electoral fates of candidates.

Diermeier and Feddersen (1998) provide insight on how legislative party discipline can arise in a parliamentary system, even without electoral party discipline. A prominent feature of parliamentary systems is the vote of confidence procedure. A governing coalition can attach a vote of confidence to any legislation so that a failure of passage induces a dissolution of the governing coalition. Since members of the governing coalition have a better chance to be recognized as agenda setters, and therefore to extract more rents, members of the governing coalition have greater continuation value under the current government than non-members do. Thus, they have more incentive to vote with the proposed rent distribution and sustain the current government. Therefore, legislative party discipline arises and, as a consequence, members of the governing coalition are able to extract more rents than in non-parliamentary settings.

On the other hand, if parties have strong control over the electoral fate of representatives, they are able to impose legislative party discipline over their members. In this case, parties or factions may act in unity, and legislative bargaining is likely to happen between leaders of parties or factions. Intra-party or intra-faction bargaining then decides how much a representative receives in rents. It is unclear in this case whether a typical member of the governing coalition would receive more rents. In the extreme case where party leaders retain all rents, having an extra representative may not increase the public spending in a district.

Though some argue that Japanese politics has become more party-centered since the electoral reform in 1994 (e.g., Rosenbluth et al., 2010), Japanese politics has traditionally been personalistic, in that personal characteristics of candidates and personal votes are much more salient than party platforms in elections. However, legislative party discipline has been strict (Hirano et al., 2011). Only in rare occasions, is the ruling coalition unable to rely on votes from its members to pass legislation. One example in which party discipline failed was the privatization reform of Japan Post in 2005, after which Prime Minister Junichiro Koizumi expelled rebels from his party and called an early election (Nemoto et al., 2008). Japan has a parliamentary system with a vote of confidence procedure. It also has strict campaign finance laws favoring parties over individual politicians. For example, public funds subsidizing political campaigns are available to qualified parties, but not to individual candidates. Individual politicians cannot legally accept campaign contributions from corporations, labor unions, and other organizations. Therefore, it is unclear to what extent the legislative party discipline in Japan can be attributed to the endogenous bargaining cohesion emphasized in Diermeier and Feddersen (1998), or to the presence of electoral institutions advantaging parties.

To see whether having a representative in the governing coalition increases municipal spending, I extend the baseline specifications by adding two dummy variables. One dummy variable equals one if the SMD representative belongs the governing coalition and zero otherwise. Another dummy variable equals one if a municipality has a PR representative belonging to the governing coalition and zero otherwise. Results using the quasi-randomized sample are reported in Table 12. Having an SMD representative in the governing coalition increases per capita municipal expenditure by about 1.5% to 1.9%, which is significant at 10%. Having a PR representative in the governing coalition increases per capita municipal expenditure more than having a non-governing PR representative, by about 0.7% to 1.4%, although the estimates are not always significant at a 10% level. This result is consistent with Albouy's (2013) findings for the U.S. However, even if the PR representative is not from the governing coalition, having a PR representative still increases per-capita municipal public expenditure by about 1.5%, which suggests that a better chance of having a candidate in the governing coalition is not the main driver of my baseline results.

The above results suggest that representatives are able to share some rents obtained by their parties or factions. In this case, the rents obtained by SMD representatives and by PR representatives should be positively correlated if they are from the same party. Since the bargaining power of a party could change from year to year even within a parliamentary term, I test the above prediction by comparing the party-year fixed effects of SMD representatives and party-year fixed effects of PR representatives. In particular, I estimate the following specification using the quasi-randomized sample:

$$log(y_{it}) = \alpha + \delta_{pt}^{SMD} + \delta_{pt}^{PR} + X'_{it}\beta + \mu_i + \pi_t + \epsilon_{it}$$

$$\tag{2}$$

where  $y_{it}$  is the public expenditure per capita for municipality *i* in fiscal year *t*;  $\delta_{pt}^{SMD}$  is a party-year fixed effect for municipality *i* with an SMD representative from party *p* in fiscal year *t*;  $\delta_{pt}^{PR}$  is a party-year fixed effect for municipality *i* with a PR representative from party *p* in fiscal year *t*, which is equal to zero if municipality *i* does not have a PR representative;  $X_{it}$  is a vector of demographic and economic controls included in column (6) of Table 8;  $\mu_i$  is a municipal fixed effect and  $\pi_t$  is a year fixed effect. The specification is estimated using dummy variables indicating party-year of the SMD representative and the PR representative (if any).

Figure 5 plots the party-year fixed effects of PR representatives  $\delta_{pt}^{PR}$  against the party-year fixed effects of SMD representatives  $\delta_{pt}^{SMD}$ . In this figure, I limit my attention to the two major parties, the Liberal Democratic Party (LDP) and the Democratic Party of Japan (DPJ), because the other parties have small numbers of SMD representatives and PR representatives in a particular year. The estimated party-year fixed effects  $\delta_{pt}^{PR}$  and  $\delta_{pt}^{SMD}$  are positively correlated. This suggests that, even where party discipline prompts legislators to vote along the party line, an extra representative is able to provide greater public expenditure by sharing rents obtained by his or her party.

## 6 Within-District Heterogeneity

In the classical Hotelling–Downs paradigm (Hotelling, 1929; Downs, 1957), the platforms of two candidates competing for office converge at the position most preferred by the median voter in a

one-dimensional policy space. With a multi-dimensional policy space, however, policy convergence is less warranted (Krasa and Polborn, 2012). Empirical observation often suggests non-convergence of policy choices by politicians. Using detail referendum voting records in Los Angeles County to measure voter preferences, Gerber and Lewis (2004) find that, in more heterogeneous districts, the voting records of legislators are more distant from the positions preferred by the median voters. In other words, electoral competition is a weaker force for heterogeneous districts in driving convergence in policy choices. Lee et al. (2004) find that the electoral strength of members of the U.S. House of Representatives explains little of the variation in their voting records. These findings suggest that voters affect policies primarily by electing representatives with fixed preferences, rather than by using elections to pressure representatives to adopt their preferred positions. In such a citizen–candidate framework (Osborne and Slivinski, 1996; Besley and Coate, 1997), some segments of voters have preferences that are more aligned with the preference of their representatives than other voters, and hence there voters are better represented. Overall, this literature suggests that within-district heterogeneity affects how a district is represented.

In the natural experiment I study here, if a district has a PR representative, the PR representative would be from a party different from that of the SMD representative. This is because parties do not nominate more than one candidate to compete for an SMD seat. The losing candidate and winning candidate are necessarily from two different parties.<sup>10</sup> This fact has two implications. First, when the runner-up of an SMD race is elected to a PR seat, voters with preferences closer to the runner-up than to the SMD winner are better represented. The gain in the effective preference representation from having a PR representative is potentially larger for heterogeneous districts. To estimate whether within-district heterogeneity affects the impact of PR representation on municipal spending, I proxy for within-district heterogeneity using the within-district standard deviations across municipalities of the municipal demographic controls used in the main estimations (i.e., population, income and age profile), as well as the share of local tax revenue in total municipal public expenditure. I then normalize these measures of within-district heterogeneity to have mean zero and standard deviation of one, and interact them with the dummy variable for having a PR representative or not. The use of cross-municipality differences to measure within-district heterogeneity is justified on two grounds. First, it may not be feasible for legislators to target groups within a

<sup>&</sup>lt;sup>10</sup>Candidates may also run as independents for an SMD seat.

municipality, while it may be feasible to target municipalities in this case, within-district crossmunicipality heterogeneity is first order important. Second, in the presence of Tiebout sorting, within-district, cross-municipality differences are positively correlated with district heterogeneity in preferences.

I find that the effect of having a PR representative on municipal expenditure is larger in districts that are more heterogeneous among the municipalities they contain. As reported in Table 13, measures of within-district heterogeneity, except for per-capita income, have small and insignificant direct effects on per-capita municipal expenditure. However, their interactions with the treatment dummy for PR representation are always positive, and all but the interactions with log income per capita and the share of local tax revenue in expenditure are significant at 1%, where the interaction with the share of local tax revenue, is significant at 10%. Moreover, the magnitudes of these interactions are large. For example, the treatment effect almost doubles in districts that are one standard deviation higher than the national average in the heterogeneity measure for the share of population aged 65 years or more. In unreported results, when the treatment dummy of having a PR representative is interacted with these municipal control variables, the interaction terms have small coefficients and are never significant. Thus, these results are not mechanically driven by any heterogeneous treatment effect along these demographic variables.

However, it is unlikely that representatives distribute public spending to their constituents purely according to their own preferences, ignoring their electoral situation. The second implication of the partisan difference between the SMD and PR representatives is that having a PR representative weakens the incumbency advantage of the SMD representative in the following election. Incumbents enjoy electoral advantages over challengers for reasons such as better name recognition among voters and access to pork barrel spending as an electoral instrument. When the challenger to an SMD incumbent is a PR representative, the SMD incumbent no longer has these advantages and hence is subject to greater electoral competition. Using the quasi-randomized sample and a linear probability model, I estimate the effects of the runners-up's election to a PR seat on the electoral performance of both SMD winners and runners-up in the following election. I find that when a runner-up in an SMD race is elected to a PR seat, she is about 50% more likely to run again in the following election, 70% more likely to be elected to an SMD seat and 50% more likely to be elected to any seat. These effects are significant at 1%. On the other hand, the winner of the SMD race is about 6% less likely to run again in the following election and about 15% less likely to be re-elected to an SMD seat. These effects are marginally significant at 10%. Reported in Table 14, these findings suggest that being a PR representative gives a runner-up incumbency advantage, which intensifies the electoral competition between the SMD winner and the runner-up.

The analysis above suggests that having a PR representative might change the electoral incentives of candidates and might differentially affect municipalities within a district. In the following section, I examine the differential impacts of having a PR representative according to the municipalities' level of electoral support for the two representatives.

## 7 Targeting: Swing versus Core

In empirical studies measuring representation as the ratio between the number of representatives and the size of the electorate, an implicit assumption is that the quantity of representatives is firstorder important. However, the results in the last section suggest that the quality of representation is at least as important. Since parties do not nominate more than one candidate in an SMD race, a PR-elected candidate will be from a party other than that of the SMD winner. Voters who preferred for the PR-elected candidate in the SMD race generally have interests different from those who voted for the SMD winner. A candidate losing the SMD race but elected through the party list affords her supporters representation in the legislature. The additional representation brings higher spending to the district, the more so in more heterogeneous districts. In other words, representation matters, but how much it matters may depend on how heterogeneous the district is.

However, voters' interests are likely both economic and ideological. Ideological affinity with a representative does not necessarily imply favorable treatments in the distributive policies decided by an office-seeking politician, specially for policies regarding tactical (pork barrel) spending. Theoretical arguments have been made on both sides about whether politicians will allocate tactical spending to swing voters or to their core supporters (e.g., Lindbeck and Weibull, 1993; Cox and McCubbins, 1986). In a general framework, Dixit and Londregan (1996) model how two parties compete for vote shares by promising pork barrel spending to groups with different partian affinity. When the two parties are symmetric in their ability to deliver pork spending to different groups, the parties will target groups containing a large share of swing voters (i.e., voters with weak predisposition toward a particular party). However, when parties are more efficient in delivering pork spending to their core supporters, perhaps due to their better understanding of what kind of public goods their core supporters want, spending is tilted toward groups with a large number of core supporters for the respective party. Existing evidence tends to support the notation that spending is targeted toward swing voters. For example, Arulampalam et al. (2009) find that Indian states that were both swing and aligned with the central government received more grants. Dahlberg and Johansson (2002) find that the central government of Sweden is more likely to provide temporary grants to municipalities with a large fraction of swing voters.

In order to examine whether the core supporters of an incumbent legislator receive more targeted transfers, I measure the swingness of a municipality by the vote share difference between the SMD winner and the runner-up in that municipality in the last election. While politicians may want in principle to target swing voters in a municipality overwhelmingly voting for one candidate, the non-partisan politics at the local level make such targeting difficult to carry out in practice. I will focus on the quasi-randomized sample. In this sample, the PR-elected candidates are almost always the runners-up, making the vote share margins in municipalities comparable between the treatment group and control group. In the eight cases where the PR-elected candidates was not the runners-up, most were closely third. Dropping these observations has little effect on the estimates. I extend the baseline specification to include a quadratic polynomial of this municipal vote share margin and its interaction with the dummy variable for PR representation as follows:

$$log(y_{it}) = \alpha + \left(\lambda_1 M M_{it} + \lambda_2 M M_{it}^2\right) + P R_{it} \times \left(\delta_0 + \delta_1 M M_{it} + \delta_2 M M_{it}^2\right) + X_{it}' \beta + \mu_i + \pi_t + \epsilon_{it} \quad (3)$$

where as before  $y_{it}$  is public expenditure per capita for municipality *i* in fiscal year *t*;  $PR_{it}$  is a dummy variable equal to one if municipality *i* has an SMD-losing but PR-elected representative at time *t* and zero otherwise;  $X_{it}$  is a vector of controls;  $\mu_i$  is a municipal fixed effect;  $\pi_t$  is a year fixed effect; and  $MM_{it}$  is the difference of vote share between the SMD winner and runner-up in the municipality in the last election (hereafter municipal margin). Unlike the victory margin in the whole district, this municipal margin can be positive or negative. A negative municipal margin indicates a stronghold for the runner-up. If spending is targeted to swing voters, I expect  $\lambda_2 < 0$  and  $\lambda_2 + \delta_2 < 0$ . On the other hand, if spendings are targeted to core supporters, I expect  $\lambda_1 M M_{it} + \lambda_2 M M_{it}^2$  to be monotonically increasing over the theoretical support of  $M M_{it}$ , i.e. [-1, 1]. Moreover, the quadratic function of  $M M_{it}$  when a municipality has a PR representative should have a U-shape, i.e.,

$$\lambda_2 + \delta_2 > 0$$
$$-1 < -\frac{\lambda_1 + \delta_1}{\lambda_2 + \delta_2} < 1$$

In order to make sure that the municipal vote share margin is not capturing a nonlinear effect of the district-wise margin of victory for the SMD winner, I include a quadratic polynomial of the district-wide margin in one specification. In another specification, I control for both the vote share of the SMD winner and that of the SMD runner-up, in case the victory margin is inadequate for measuring the electoral safety of the nominal representative. As reported in the Table 15, none of these controls is significant in predicting municipal expenditures. In the first three columns of Table 15, I first report results of regressions restricting all coefficients of quadratic terms to zero. In municipalities with no PR representative, the municipal margin has positive but small and insignificant effects on municipal expenditure. Having a PR representative has little impact on the effects of the municipal margin. In the last three columns of Table 15, the regression results suggest no apparent nonlinear relationship between municipal margin and municipal expenditure when there is no PR representative. However, when having a PR representative, there is a significant quadratic relationship between municipal margin and municipal expenditure. This result suggests that municipalities that vote heavily for either the SMD winner or the runner-up have more spending. In municipalities where the SMD winner and runner-up obtain equal vote shares, having a PR representative increases municipal expenditure by 1%. However, in municipalities where the runner-up outperforms the winner by 10 percentage points, having a PR representative increases expenditure by 2.3%.

In the top-left panel of Figure 6, I plot the fitted quadratic relationship between municipal margin and municipal expenditure, for the cases in which the municipality does and does not have a PR representative. Controls and fixed effects are set to zero for this graph. The kernel density of the municipal margin is in the background. Moreover, I set the dependent variable to be transfers from the central government or public works expenditure, respectively, both in log per capita terms.

The specifications are otherwise the same as in Eq. (2) and column (4) of Table 15. Alternative choices of controls as listed in Table 15 have little impact on the relationship between municipal margin and the dependent variable. The regression results are plotted in the middle and right graphs on the top row of Figure 6. The patterns across these three graphs are similar. Having a PR representative increases spending and transfers in the municipalities where the runner-up did well. To examine whether this result is due to a misspecification of quadratic polynomial, I replace the quadratic polynomial with a cubic spline with internal knots at the 25th, 50th, and 75th percentiles of the municipal margin. The regression results for municipal expenditure, transfers and public works spending (all in log per capita terms) respectively are plotted in the bottom row of Figure 6. The cubic results confirm the bipolar effects of having a PR representative.

Interestingly, having a PR representative increases spending not only in municipalities where the PR representative had strong support, but also in municipalities where the SMD winner had a large lead. In other words, the presence of a PR representative shifts distributive politics toward core targeting. Without a PR representative, core targeting is mild, but the presence of a PR representative intensifies core targeting. This result suggests that incumbents do not provide pork simply to carry out promises made when they ran for election; rather, they distribute pork in reaction to their electoral situation.

My interpretation is that politicians distribute pork barrel spending to mobilize voters. Maybe except in Drazen and Eslava (2012), models of distributives politics tend to abstract from the turnout decisions of voters. Taking voter turnout as given, politicians distribute pork barrel spending to groups containing a large fraction of swing voters because the spending could switch a large number of votes. If the two candidates are symmetric and both able to credibly promise an allocation of spending, this vote-buying incentive favors swing groups in distributive politics (Dixit and Londregan, 1996). However, only incumbent politicians may affect the allocation of pork barrel spending and turnout decisions of voters may depend on the allocation of spending as voting is costly. In Japan's setting, when there is only one incumbent, the vote-buying incentive would induce him to distribute benefit toward municipalities with a lot of swing voters, while the voter mobilization or turnout-buying incentive would induce him to distribute benefits toward municipalities with a lot of supporters. Combining these effects, a sole incumbent favors neither swing municipalities nor core municipalities. Hence, as shown in the data, the correlation between the incumbent's vote share in a municipality and that municipality's public expenditure is positive but weak statistically and in magnitude. From Figure 6, when there is no PR representative, swing municipalities have spending levels similar to municipalities that voted strongly for the SMD winner.

However, when there are two incumbents who are able to distribute pork, the mobilization incentive dominates. This could happen due to an inference problem for voters to assign credit to politicians bringing in spending. Incumbents may target swing municipalities and their core municipalities. When there are two incumbents, voters in swing municipalities may have difficulties to identify the contribution of each incumbent to the increase of public spending. Thus, the electoral return from distributing pork toward core municipalities becomes higher relative to the benefit of distributing toward swing municipalities.

Some additional empirical evidence supports such interpretation. I estimate a specification similar to that in Eq. (3) but with the turnout rate in the next election as the dependent variable. The municipal turnout rate is estimated by using a municipality's total number of valid votes cast divided by the voting age population. When there are two incumbents, the turnout rates in the next election are higher, particularly in municipalities that strongly favor either the SMD winner or the runner-up. The quadratic relationship is plotted in Figure 7. A cubic spline similar to those used in Figure 6 also confirms a bipolar impact of having a PR representative on turnout. However, possibly due to a smaller sample size and measurement errors in the turnout rate, the relationship between turnout rate and a representative's strength of support in a municipality is less precisely estimated.

When there are two incumbents, strongholds of either incumbent benefit from greater local public expenditure. I argue that the intensified electoral competition provides greater incentive for politicians to bring more public spending to their core constituents. However, an alternative explanation is that electoral competition induces the majority party or the governing coalition to distribute more public spending to their core constituents. The governing coalition does so to protect their SMD seats in districts they win or to contest for the SMD seats in districts they lose. Because in some districts, the candidates from the governing coalition are the SMD winners while in other districts, the candidates from the governing coalition are runners-up, distributing more public spending to governing coalition's strongholds may drive a bipolar relationship between the margin of vote share between SMD winners and runner-up, as found previously. To assess this possibility, I replace the municipal margin of vote share between the SMD winner and runner-up with the municipal margin of vote shares between the candidate from the governing coalition with the highest vote share in the district and the candidate from an opposition party with the highest votes share in the district. Then, I re-run the regressions shown in Figure 6. If the bipolar relationship found in Figure 6 is driven by the governing coalition, there should not be a bipolar relationship between this new municipal margin and public spending. In particular, having an additional representative should only benefit municipalities where the governing coalition has strong support. The new set of regression results is plotted in Figure 8 in a way similar to Figure 6. As shown in the top left plot of Figure 8, there is a similar bipolar relationship between the new municipal margin and the total municipal expenditure when there are two incumbents. Not only the strongholds of the governing coalition, which are located on the right side of the plot, benefit from having an additional representative, so do the strongholds of the opposition parties, which are located on the left side of the plot. Similar relationships are found when the dependent variable is transfers from the central government or infrastructure spending, and when the polynomial fit is quadratic or cubic spline.

The analysis above supports that electoral competition incentivizes individual politicians in the quasi-randomized sample to bring in more public resources to their core constituents. However, this is not to say parties do not react to electoral situations. The analysis in this section may not be best suitable for an analysis of electoral competition among parties. From the perspective of parties, they concern about the number of SMD seats they win and the vote shares they have in the PR ballots. However, the most competitive SMD races are not likely to be included in the quasi-randomized sample. This is because the dual candidates who narrowly lose in SMD races are likely to rank very high on her party lists. A small vote share shock may be insufficient to take away the PR seat of a losing candidate, or if it changes the result of a SMD race, whoever loses in the SMD race would have a high rank on her party list, assuring her district two representatives. Moreover, the competition among parties for PR votes in the (larger) PR blocks may not be well captured in the analysis in this section. As illustrated in Persson and Tabellini (2000) and Milesi-Ferretti et al (2002), geographically targeted spending may be less effective as an instrument for electoral competition when the electoral system is based on the principle of proportional representation rather than winner-take-all.

The analysis in this section adds to the literature of distributive politics by highlighting the important distributional implications of whether the electoral competition is among individual politicians or among parties. As summarized in Golden and Min (2013), theories are ambiguous about whether swing voters or core voters would be targeted in distributive politics and empirical evidence is mixed. In the influential works of Lindbeck and Weibull (1987), Cox and McCubbins (1986), and Dixit and Londregan (1996), parties and / or candidates are assumed to be able to make credible commitments about their proposal distribution of spending or transfers, and there is no agency problem between parties / candidates and voters. The analysis in the section suggests that, when the primary actors in the electoral competition are individual incumbents, it may not be appropriate to abstract from the commitment problems and agency problems for the analysis of distributive politics.

# 8 Concluding Remarks

The empirical results in this paper suggest that political representation affects the allocation of public expenditure. Having an additional incumbent legislator with electoral interests in the district increases local public expenditure, in particular discretionary spending on public works. Heterogeneous districts benefit more from this extra representation. Having a second representative elected to the legislature intensifies subsequent electoral competition; nevertheless, this heightened competition does not generate policy convergence, at least not in distributive policies toward local governments. Instead, core supporters of both representatives in the district receive a higher amount of transfers from the central government. Turnout-buying may partially explain why swing voters are not targeted, despite contrary theoretical arguments favoring targeting of swing voters. To obtain votes, politicians can either attract votes from other candidates or they can turn out voters likely to vote for them. The latter motive favors targeting core supporters in tactical spending.

Partisan representation affects programmatic redistributions over the median term. In the U.S., Levitt and Snyder (1995) find that federal outlays are higher in districts with a large number of Democratic voters, especially for programs that were initiated when the Democratic Party had a large majority in Congress 10 to 15 years before the outlays and that are currently administrated based on pre-established formulas. Brender and Drazen (2013) find that replacement of political leaders affects the composition of public expenditures in the medium term, but not in the short term. This paper adds to the literature by providing evidence that having extra representation could also provide short-term discretionary spending benefit to voters with strong partisan or ideological affinity with the added representative. This result suggests that the heterogeneity of an electoral district is important for distributive policies, which has policy implications for redistricting and the design of electoral systems.

Moreover, this paper documents a subtlety in Japan's mixed-member electoral system that makes representatives elected through party lists responsive to geographically narrow interests. A large literature debates the relative merits of two electoral systems: a Majoritarian system, such as in the U.S., and a Proportional Representation system, as is common in Europe. Recent studies emphasize the positive implications of the electoral system for public finance. Notably, Melesi-Ferretti et al. (2002) argue that, relative to a PR system, a Majoritarian system tends to spend more on goods and services *vis-à-vis* transfers. The rationale is that spending in goods and purchases is easier to target toward geographic areas, while transfers better target socio-economic groups. A middle ground may seem to be a mixed Majoritarian–PR system, which is often thought to have the best of two worlds and has been adopted by a number of new and existing democracies in the recent decades, such as Taiwan and New Zealand. In this paper, I show that an electoral feature - how PR representatives are ranked on their party lists - compromises the supposedly broad representation by PR representatives in a mixed system.

# References

- Ackerman, Bruce (2000), "The New Separation of Powers," *Harvard Law Review*, Vol. 113, No. 3 (Jan., 2000), pp. 633-729
- Rosenbluth, Frances, Jun Saito, and Kyohei Yamada (2010), "Electoral Adaptation in Japan: Party Strategy after Electoral Rule Change," Department of Political Science, Yale University.
- [3] Albouy, David (2013), "Partisan Representation in Congress and the Geographic Distribution of Federal Funds," *Review of Economics and Statistics* 95, no. 1: 127-141.
- [4] Alesina, Alberto, Reza Baqir and William Easterly (1999), "Public Goods and Ethnic Divisions," The Quarterly Journal of Economics, November 1999.
- [5] Alesina, Alberto, and Eliana La Ferrara (2000), "Participation in Heterogeneous Communities," *The Quarterly Journal of Economics* 115, no. 3 : 847-904.
- [6] Ansolabehere, Stephen A., Alan Gerber and Jim Snyder (2002), "Equal Votes, Equal Money: Court-Ordered Redistricting and Public Expenditures in the American States," American Political Science Review, Vol. 96, No. 4 December 2002.
- [7] Atlas, Cary M., Thomas W. Gilligan, Robert J. Hendershott, and Mark A. Zupan (1995),
   "Slicing the Federal Government Net Spending Pie: Who Wins, Who Loses, and Why," *The American Economic Review* 85, no. 3: 624-629.
- [8] Arulampalam, Wiji, Sugato Dasgupta, Amrita Dhillon, and Bhaskar Dutta (2009), "Electoral Goals and Center-State Transfers: A Theoretical Model and Empirical Evidence from India," *Journal of Development Economics* 88, no. 1: 103-119.
- Besley, Timothy, and Robin Burgess (2002), "The Political Economy of Government Responsiveness: Theory and Evidence from India," *The Quarterly Journal of Economics* 117, no. 4: 1415-1451.
- [10] Besley, Timothy, and Stephen Coate (1997), "An Economic Model of Representative Democracy," The Quarterly Journal of Economics 112, no. 1: 85-114.

- [11] Baron, David P., and John A. Ferejohn (1989), "Bargaining in Legislatures," The American Political Science Review : 1181-1206.
- [12] Besley, Timothy, and Stephen Coate (2003), "Centralized versus Decentralized Provision of Local Public Goods: a Political Economy Approach," *Journal of Public Economics* 87, no. 12: 2611-2637.
- [13] Besley, Timothy (2007), "Principled Agents?: The Political Economy of Good Government," Oxford University Press.
- [14] Brender, Adi, and Allan Drazen (2012,)"Elections, Leaders, and the Composition of Government Spending," *Journal of Public Economics*.
- [15] Cameron, A. Colin and Pravin Trivedi (2005), "Microeconometrics: Methods and Applications," *Cambridge University Press*, New York, May 2005.
- [16] Dahlberg, Matz, and Eva Johansson (2002), "On the Vote-Purchasing Behavior of Incumbent Government," American Political Science Review 96, no. 1: 27-40.
- [17] DeWit, Andrew (2002), "Dry Rot: The Corruption of General Subsidies in Japan," Journal of the Asia Pacific Economy 7, no. 3: 355-378.
- [18] Downs, Anthony (1957), "An Economic Theory of Democracy," 260-276.
- [19] Drazen, Allan and Marcela Eslava (2012), "Programmatic Targeting," Working paper.
- [20] Epple, Dennis, and Richard E. Romano (1996), "Ends against the middle: Determining Public Service Provision When There Are Private Alternatives," *Journal of Public Economics* 62, no. 3 : 297-325.
- [21] Shugart, Matthew, and Martin P. Wattenberg, eds (2001), "Mixed-Member Electoral Systems: The Best of Both Worlds?: The Best of Both Worlds?," Oxford University Press.
- [22] Folke, Olle (forthcoming), "Shades of Brown and Green: Party Effects in Proportional Election Systems," Journal of the European Economic Association
- [23] Friedman, John N., and Richard T. Holden (2008). "Optimal Gerrymandering: Sometimes Pack, but Never Crack," American Economic Review, 98(1): 113-44.

- [24] Golden, Miriam, and Brian Min (2013), "Distributive Politics Around the World," Annual Review of Political Science 16: 73-99.
- [25] Krasa, Stefan, and Mattias K. Polborn (2012), "Political Competition Between Differentiated Candidates," *Games and Economic Behavior* 76, no. 1: 249-271.
- [26] Horiuchi, Yusaku and Jun Saito (2003), "Reapportionment and Redistribution: Consequence of Electoral Reform in Japan," *American Journal of Political Science*, Vol. 47, No.4, October 2003, Pp. 669-682.
- [27] Hotelling, Harold (1929), "Stability in Competition," The Economic Journal 39, no. 153: 41-57.
- [28] Gerber, Elisabeth R., and Jeffrey B. Lewis (2004), "Beyond the Median: Voter Preferences, District Heterogeneity, and Political Representation," *Journal of Political Economy* 112, no. 6: 1364-1383.
- [29] Groseclose, Tim, and James M. Snyder Jr (1996), "Buying Supermajorities," American Political Science Review: 303-315.
- [30] Milesi-Ferretti, Gian Maria, Roberto Perotti and Massimo Rostagno (2002), "Electoral Systems and Public Spending," The Quarterly Journal of Economics (2002) 117 (2): 609-657.
- [31] Myerson, Roger B. (1997), "Economic Analysis of Political Institutions: An Introduction," in D. M. Kreps and K. F. Wallis: Advances in Economics and Econometrics: Theory and Applications: Seventh World Congress. Cambridge; New York and Melbourne: Cambridge University Press, 1997, pp. 46-65
- [32] Lee, David, Enrico Moretti and Matthew J. Butler (2004) "Do Voters Affect or Elect Policies? Evidence from the U.S. House," *Quarterly Journal of Economics*, 119(3), 807-859.
- [33] Lee, Frances E. (1998), "Representation and Public Policy: The Consequences of Senate Apportionment for the Geographic Distribution of Federal Funds," *The Journal of Politics*, Volume 60 / Issue 01 / February 1998, pp 34-62.

- [34] Levitt, Steven D. and James M. Snyder (1995), "Political Parties and the Distribution of Federal Outlays," American Journal of Political Science, 39: 958–980.
- [35] Pettersson-Lidbom, Per (2008), "Do Parties Matter for Economic Outcomes? A Regression Discontinuity Approach," Journal of the European Economic Association 6, no. 5: 1037-1056.
- [36] Lizzeri, A., and Nicola Persico (2001), "The Provision of Public Goods under Alternative Electoral Incentives," American Economic Review, 225-239.
- [37] Lockwood, Ben (2002), "Distributive Politics and the Costs of Centralization," The Review of Economic Studies 69, no. 2: 313-337.
- [38] Mckean, Margaret and Ethan Scheiner (2000), "Japan's New Electoral System: la plus ca change ...," *Electoral Studies*, 19 (2000) 447-477.
- [39] Nemoto, Kuniaki, Ellis Krauss, and Robert Pekkanen (2008), "Policy Dissension and Party Discipline: the July 2005 Vote on Postal Privatization in Japan," *British Journal of Political Science* 38, no. 3: 499.
- [40] Osborne, Martin J., and Al Slivinski (1996), "A Model of Political Competition with Citizen-Candidates," *The Quarterly Journal of Economics* 111, no. 1: 65-96.
- [41] Persson, Torsten, Guido Tabellini, and Francesco Trebbi (2003), "Electoral Rules and Corruption," Journal of the European Economic Association 1, no. 4: 958-989.
- [42] Persson, Torsten, and Guido Enrico Tabellini (2000) Political Economics: Explaining Economic Policy, The MIT Press.
- [43] Persson, Torsten, and Guido Enrico Tabellini (2003), The Economic Effects of Constitutions, The MIT Press.
- [44] Schlesinger, J. M. (1999), "Shadow Shoguns: The Rise and Fall of Japan's Postwar Political Machine," Stanford University Press.
- [45] Taniguchi, Naoko (2008), "Diet Members and Seat Inheritance: Keeping It in the Family," Democratic Reform in Japan: Assessing the Impact, edited by Sherry L. Martin and Gill Steel, 2008.

- [46] Tiebout, Charles M. (1956), "A Pure Theory of Local Expenditures." The Journal of Political Economy 64, no. 5 : 416-424.
- [47] Trebbi, Francesco, Philippe Aghion, and Alberto Alesina (2008), "Electoral rules and minority representation in US cities," *The Quarterly Journal of Economics* 123, no. 1 (2008): 325-357.
- [48] Weese, Eric (2012), "Efficiencies of Scale in Local Public Good Provision: Estimation via Coalition Panel Data," Working Paper, Yale University.



Figure 1: Public Expenditure by National and Subnational Governments in Japan and U.S.

Notes: Defense expenditures are excluded for both Japan and the U.S. Total expenditures by subnational governments are equal to locally financed expenditure plus transfers from the national government.



Figure 2: Average Revenue Sources of Municipal Governments



Figure 3: Median Revenue Sources of Municipal Governments



Figure 4: The Division of PR Blocks and the Number of SMDs in Each Prefecture

Figure 5: Party-Year Fixed Effects of SMD Representatives and PR Representatives on Municipal Expenditure



The above graph plots the party-year fixed effects of PR representatives against the party-year fixed effects of SMD representatives. Party-year fixed effects are estimated from the following specification using the quasi-randomized sample described in Section 3:

$$log(y_{it}) = \alpha + \delta_{pt}^{SMD} + \delta_{pt}^{PR} + X_{it}^{\prime}\beta + \mu_i + \pi_t + \epsilon_{it}$$
<sup>(2)</sup>

where  $y_{it}$  is the public expenditure per capita for municipality *i* in fiscal year *t*;  $\delta_{pt}^{SMD}$  is a party-year fixed effect for municipality *i* with an SMD representative from party p in fiscal year t;  $\delta_{pt}^{PR}$  is a party-year fixed effect for municipality i with a PR representative from party p in fiscal year t, which is equal to zero if municipality i does not have a PR representative;  $X_{it}$  is a vector of demographic and economic controls included in column (6) of Table 8;  $\mu_i$  is a municipal fixed effect and  $\pi_t$  is a year fixed effect. The specification is estimated using dummy variables indicating party-year of SMD representative and PR representative (if any).



The dependent variables are total municipal expenditure (left column); amount of transfers from the national government (middle column); and municipal expenditure on public works (right column). All are in log per capita terms. The upper row plots the fitted values of the dependent variable against a quadratic polynomial of the municipal margin Margin with 3 internal knots at the 25th, 50th, and 75th percentiles. The dashed lines plot the kernel density of Municipal Margin. All regressions use the quasi-randomized of vote shares between the SMD winner and runner-up (Municipal Margin). The lower row plots the fitted values of the dependent variables against a cubic spline of Municipal sample and include municipality and year fixed effects. Explanatory variables other than the polynomial or spline of Municipal Margin are all set to zero.





election against a quadratic polynomial of the municipal margin of vote shares between the SMD winner and runner-up (Municipal Margin). The right graph plots The dependent variable is the turnout rate of the municipality in the following election. The left graph plots the fitted values of the turnout rate in the next the fitted values of the turnout rate in the next election against a cubic spline of Municipal Margin with 3 internal knots at the 25th, 50th, and 75th percentiles. The dashed lines plot the kernel density of Municipal Margin. All regressions use the quasi-randomized sample and include municipality and election fixed effects. Explanatory variables other than the polynomial or spline of Municipal Margin are all set to zero. Figure 8: Municipal Vote Share Margin of Governing Party Candidate over Opposition Party Candidate, PR Representation, and Expenditures and Transfers



The dependent variables are total municipal expenditure (left column); amount of transfers from the national government (middle column); and municipal expenditure on All are in log per capita terms. The upper row plots the fitted values of the dependent variable against a quadratic polynomial of the municipal vote shares margin between the governing party candidate with the highest vote share in the district and the opposition party candidate with highest vote share in the district (Municipal Margin of Gov. Party Candidate or MMG). The lower row plots the fitted values of the dependent variables against a cubic spline of MMG with 3 internal knots The dashed lines plot the kernel density of MMG. All regressions use the quasi-randomized sample and include municipality and year fixed effects. Explanatory variables other than the polynomial or spline of MMG are all set to zero. at the 25th, 50th, and 75th percentiles. public works (right column).

	lable I: Division	of Pro	portion	al Representati	on Blocks	
			# PR	Seats	# SN	MDs
PR Block	# Prefectures	1996	2000	2003 - 2012	Before 2002	After 2002
Chugoku	5	13	11	11	21	20
Hokkaido	1	9	8	8	13	12
Hokurikushinetsu	5	13	11	11	20	20
Kinki	6	33	30	29	47	48
Kitakanto	4	21	20	20	31	32
Kyushu	8	23	21	21	38	38
Minamikanto	3	23	21	22	32	34
Shikoku	4	7	6	6	13	13
Tohoku	6	16	14	14	26	25
Tokai	4	23	21	21	34	33
Tokyo	1	19	17	17	25	25
Total	47	200	180	180	300	300

 Table 1: Division of Proportional Representation Blocks

Dependent Variable: =1 if the C	andidate Rur	is Again in the	Same SMD;	0 otherwise
	0	LS	Lo	git
	(1)	(2)	(3)	(4)
Being a PR Incumbent	$0.0862^{***}$	$0.0696^{***}$	$0.1147^{***}$	0.0978***
	(0.0147)	(0.0151)	(0.0271)	(0.0273)
Losing Margin in the Last SMD Election		$-0.1415^{***}$		$-0.1271^{***}$
		(0.0468)		(0.0446)
Constant	$0.8801^{***}$	$0.9145^{***}$		
	(0.0111)	(0.0140)		
# Observations	1215	1215	1215	1215
The sample consists of SMD-losing candidates wh	o run again as	an SMD candida	te in the next el	ection;
<b>PR Incumbent</b> is a dummy variable equal to on	e if the candid:	ate running again	_	

Table 2: Probability of an SMD-Losing Candidate Running in the Same SMD in the Next Election

is an incumbent of a PR seat; and zero otherwise;

Losing Margin is the difference of vote share between the SMD winner and the

SMD-losing candidate in the last election;

Standard errors are in parentheses;

Standard errors of the OLS estimates are heteroskedasticity robust;

Estimates from the logit models are average marginal effects;

Standard errors of the logit estimates are calculated using the Delta Method;

	Table 3: I	Electoral Advantag	ges of PR Incumber	its and SMD Incu	umbents	
		Depender	it Variable: $=1$ if F	lected to $\dots$ ; 0 o	therwise	
	Elected to	Elected to	Elected to	Elected to	Elected to	Elected to
	an SMD Seat	Any Seat	Any Seat	Any Seat	Any Seat	an SMD Seat
	(1)	(2)	(3)	(4)	(5)	(9)
SMD Incumbent	$0.2851^{***}$	$0.1634^{***}$	$0.2073^{***}$	$0.2513^{***}$	$0.0890^{***}$	$0.0890^{**}$
	(0.0204)	(0.0162)	(0.0188)	(0.0199)	(0.0026)	(0.0388)
Incumbent	$0.1292^{***}$	$0.2598^{***}$	$0.1792^{***}$	$0.1506^{***}$		
	(0.0170)	(0.0180)	(0.0255)	(0.0234)		
Constant	$0.1541^{***}$	$0.3460^{***}$	$0.3740^{***}$	$0.2561^{***}$	$0.5777^{***}$	$0.4649^{***}$
	(0.0058)	(0.0084)	(0.0113)	(0.0067)	(0.0015)	(0.0239)
Fixed Effects	Party–Election	Party–Election	Party–Election	Party–Election	Party–Election	Party-Election
Clustering	SMD–Election	Block–Election	Block–Election	Block–Election	Block–Election	SMD–Election
$\operatorname{Sample}$	SMD Candidates	<b>PR</b> Candidates	<b>Dual Candidates</b>	All Candidates	Incumbents $\mathbf{v}$	who were dual
	(incl. Dual)	(incl. Dual)	Only		candidates in clo	se SMD elections
R-Squared	0.5471	0.4092	0.4433	0.5702	0.0647	0.6330
# Obs.	5647	4432	3507	5647	356	343
Incumbent is a dun	1 nmy variable equal to o	ne if the candidate is	an incumbent; zero ot	herwise;		
SMD Incumbent is	s a dummy variable equ	al to one if the candi	date is an SMD incum	bent; zero otherwise;		

For Column (5) and (6), the sample consists of incumbents who were dual candidates in the last election, and who were

either a winner or a runner-up in SMD elections with margin of victory less than 4%.

Standard errors robust to clustering are in parentheses.

Rank		Narrowness	Rank	Seat	SMD	V. Share	Winner's
ex ante	Candidate	ratio	ex post	won	(dual)	in SMD	v. share
1	Genichiro Sata	_	1	$\mathbf{PR}$	_	_	_
$-\frac{1}{2}$	Yuko Obuchi			$\bar{SMD}$	$\overline{\text{Gunma}}$	-0.710	0.710
2	Toshimitsu Motegi	-	-	SMD	Tochigi 5	0.517	0.517
2	Hiroshi Kajiyama	-	-	SMD	Ibaraki 4	0.507	0.507
2	Fukushiro Nukaga	0.976	2	$\mathbf{PR}$	Ibaraki 2	0.479	0.491
2	Keiko Nagaoka	0.812	3	$\mathbf{PR}$	Ibaraki 7	0.301	0.370
2	Yoshitaka Shindo	0.801	4	$\mathbf{PR}$	Saitama 2	0.401	0.500
2	Masahiko Shibayama	0.797	5	$\mathbf{PR}$	Saitama 8	0.391	0.491
2	Tsutomu Sato	0.781	6	$\mathbf{PR}$	Tochigi 4	0.402	0.515
2	Yuya Niwa	$0.772^{}$	7		Ībarakī 6	$-\bar{0.420}$	0.543
2	Yasuaki Yamaguchi	0.770	8	-	Saitama 10	0.425	0.551
2	Hajime Funada	0.765	9	-	Tochigi 1	0.413	0.540
2	Taku Otsuka	0.728	10	-	Saitama 9	0.412	0.567
2	Toshio Kojima	0.715	11	-	Saitama $12$	0.409	0.572
2	Yoshio Tanaka	0.711	12	-	Saitama 15	0.371	0.521
2	Yasuhiro Hanashi	0.702	13	-	Ibaraki 3	0.401	0.571
2	Shinako Tsuchiya	0.701	14	-	Saitama $13$	0.361	0.515
2	Takashi Mitsubayashi	0.687	15	-	Saitama 14	0.393	0.572
2	Hideki Makihara	0.650	16	-	Saitama $5$	0.385	0.592
2	Hayakawa Chuko	0.618	17	-	Saitama $4$	0.335	0.542
2	Hiroshi Imai	0.617	18	-	Saitama 3	0.371	0.600
2	Norihiko Akagi	0.612	19	-	Ibaraki 1	0.350	0.571
2	Hideaki Okabe	0.594	20	-	Ibaraki 5	0.364	0.613
2	Koya Nishikawa	0.567	21	-	Tochigi 2	0.357	0.629
2	Zenjiro Kaneko	0.476	22	-	Saitama 1	0.290	0.609
2	Kazuyuki Nakane	0.453	23	-	Saitama 6	0.306	0.676
2	Etsuji Arai	0.363	24	-	Saitama 11	0.256	0.707
$\bar{28}$	Masayoshi Namiki		-25				
29	Matsuo Otaka	-	26	-	-	-	-

Table 4: Party List of LDP for the PR Block of Kitakanto in the General Election of 2009

Kitakanto is an area north of the Tokyo prefecture.

SMDs are named with its prefecture followed by the district number in the prefecture.

For example, Gunma 5 is District 5 of Gunma Prefecture;

In the general election of 2009, LDP won 25.84% of party votes in the PR Block of Kitakanto.

Therefore, 6 out of 20 seats in the PR block were allocated to LDP.

Moreover, 3 dual candidates won in their SMDs, thereby vacating their positions on the party list.

The last column is the vote share of the winner in the SMD the dual candidate is contesting.

Table 5: Distribution of SMDs w	ith PR-	-elected	Repre	sentativ	ves	
	]	Number	<b>Full S</b> r of SM	ample Ds by I	Election	1
Number of Additional PR Representatives	1996	2000	2003	2005	2009	2012
0 (without dual candidate)	12	5	3	0	0	2
0 (with dual candidate)	212	220	182	186	206	183
1	70	69	111	111	91	105
2	6	6	4	3	3	10
Total	300	300	300	300	300	300

## **Quasi-Randomized Sample** Number of SMDs by Election

	1	vumber	OI SIM	DS Dy I	Election	1
Treatment Status	1996	2000	2003	2005	2009	2012
One Additional PR Representative No Additional PR Representative	$\begin{array}{c} 24 \\ 54 \end{array}$	$\begin{array}{c} 24 \\ 69 \end{array}$	57 66	37 80	$\frac{32}{69}$	$\begin{array}{c} 37 \\ 70 \end{array}$
Total	78	93	123	117	101	107

	Univ	variate Ez	xplanatory	Variable:
	=1 if having	g an PR	Representa	tive; 0 otherwise
Dependent Variable	Coefficient	S.E.	<i>p</i> -value	# Obs.
Municipal Level:				
Population (log)	-0.0053	0.1018	0.9584	10208
Taxable Income per capita (log)	-0.0201	0.0233	0.3889	10156
Population Share of Age 0 to 4	0.0006	0.0006	0.2957	10208
Population Share of Age 5 to 19	0.0004	0.0023	0.8558	10208
Population Share of Age $65 +$	0.0007	0.0056	0.8965	10208
Area (log)	0.0310	0.1068	0.7714	9251
Population Density (log)	-0.0483	0.1739	0.7813	9251
Population Growth Rate	-0.0004	0.0013	0.7645	9336
District Level:				
Size of District Electorate (log)	-0.0153	0.0184	0.4067	512
Number of SMD Candidates	-0.0299	0.0851	0.7254	619
Total Vote Share of Top 2 SMD Candidates	0.0043	0.0100	0.6640	619
LDP Candidate won	-0.0332	0.0422	0.4313	619
DPJ Candidate won	-0.0714*	0.0395	0.0710	619
Voter Turnout Rate	0.0034	0.0063	0.5899	512
SMD Winner in Governing Coalition	-0.0542	0.0424	0.2014	619

## Table 6: Quasi-Randomization Check: Whether Treatment Status Correlates with Observables

LDP Candidate equals to 1 if the SMD seat is won by a LDP candidate; 0 otherwise;

**DPJ Candidate** equals to 1 if the SMD seat is won by a DPJ candidate; 0 otherwise;

The sample is the quasi-randomized sample constructed with 2% vote share perturbations.

Standard errors in parentheses are robust to clustering two-way on municipality and on SMD–House term. \* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.

		Dependent Varia	ble: Municipal P	ublic Expenditure	e per capita (log)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	$0.0085^{***}$ (0.0033)	$0.0085^{***}$ (0.0032)	$0.0080^{**}$ (0.0034)	$0.0082^{**}$ (0.0032)	$0.0087^{***}$ (0.0032)	$0.0073^{**}$ (0.0030)
Population (log)		0.0211 (0.0593)				0.0911 (0.0554)
Taxable Income per capita (log)			$0.1503^{***}$			$0.1942^{***}$
			(0.0513)			(0.0476)
Age 0 to $4 /$ Population				$-1.2553^{**}$		$-1.1071^{**}$
				(0.5185)		(0.5036)
Age 5 to $19 \ / \ Population$				0.0584		0.1413
				(0.3401)		(0.3457)
Age $65 + /$ Population				$1.1511^{***}$		$1.5540^{***}$
				(0.2478)		(0.2741)
Size of District Electorate (log)					-0.0124	-0.0068
					(0.0463)	(0.0464)
Fixed Effects	Municipal	Municipal	Municipal	Municipal	Municipal	Municipal
	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year
Weighted	No	$N_{O}$	$N_{O}$	No	No	$N_{O}$
R-squared	0.9596	0.9596	0.9601	0.96	0.9596	0.9608
# Obs.	34906	34906	34717	34906	34838	34649
PR Representative is a dummy varia	able equal to one if t	the municipality has	at least one PR repr	esentative and zero o	otherwise;	

Table 7: Additional Representation and Local Public Expenditure Per Capita (Full Sample)

Standard errors in parentheses are robust to clustering two-way on municipality and on PR block-House term; \* p < 0.10; \*\*\* p < 0.05; \*\*\* p < 0.01.

		Dependent Varia	ble: Municipal P	ublic Expenditure	e per capita (log)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	$(3) \\ Coeff. / (S.E.)$	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	$0.0182^{***}$ (0.0066)	$0.0178^{**}$ (0.0070)	$0.0155^{**}$ (0.0063)	$\begin{array}{c} 0.0184^{***} \\ (0.0068) \end{array}$	$\begin{array}{c} 0.0210^{***} \\ (0.0062) \end{array}$	$0.0182^{***}$ (0.0060)
Population (log)		-0.1860 (0.1223)				-0.0513 (0.1039)
Taxable Income per capita (log)			$0.2912^{***}$			$0.2866^{***}$
Age 0 to 4 / Population			(0.0814)	$-2.0947^{**}$		(0.0603) -1.4602*
				(0.9088)		(0.8786)
Age 5 to 19 $/$ Population				0.6212		$0.9935^{**}$
				(0.4195)		(0.4699)
Age $65 + /$ Population				$0.9743^{**}$		$1.3175^{***}$
				(0.3910)		(0.3831)
Size of District Electorate (log)					$-0.1914^{***}$	$-0.2000^{***}$
					(0.0579)	(0.0733)
Fixed Effects	Municipal	Municipal	Municipal	Municipal	Municipal	Municipal
	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year
Weighted	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$
R-squared	0.9717	0.9722	0.9732	0.972	0.972	0.9739
# Obs.	10208	10208	10156	10208	10184	10132
<b>PR Representative</b> is a dummy varia	able equal to one if th	ne municipality has	a PR representative	and zero otherwise;		

Table 8: Additional Representation and Local Public Expenditure per capita (Quasi-randomized Sample with 2% Perturbations)

Standard errors in parentheses are robust to clustering two-way on municipality and on PR block-House term;

Dependent Vari	iable: Transfers fi	rom National Gor	vernment to Mun	icipal Governmen	ts (log per capita	<b>(1</b> )
			Full Sample (F)	Y 2002 - FY 2010	(	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) $Coeff. / (S.E.)$
PR Representative	$0.0091^{**}$ (0.0044)	$0.0092^{**}$ (0.0043)	$0.0093^{**}$ (0.0044)	$0.0093^{**}$ (0.0044)	$0.0088^{**}$ (0.0044)	$0.0087^{*}$ (0.0045)
R-squared # Obs.	0.9788 17367	0.9789 17367	$\begin{array}{c} 0.9788\\ 17216\end{array}$	0.9792 17367	0.9788 17299	0.9793 17148
		Quasi-r	andomized San	aple (FY 2002 - I	۲ 2010)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	0.0175 (0.0107)	0.0173 (0.0106)	0.0177 (0.0108)	0.0187 (0.0118)	0.0174 (0.0109)	0.0185 (0.0119)
R-squared # Obs.	0.9857 5956	0.9857 5956	0.9857 5912	0.9858 5956	0.9857 5932	0.9858 5888
Control Variables:	(1)	(2)	(3)	(4)	(5)	(9)
Population (log) Tavable Income ner canita (loc)		Υ	>			Y
Age 0 to 4 / Population			4	Υ		۲ ۲
Age 5 to $19$ / Population				Υ		Υ
Age 65+ / Population Size of District Electorate (log)				Υ	Υ	YY
<b>PR Representative</b> is a dummy vari Standard errors in parentheses are roby	iable equal to one if t ust to clustering two-	he municipality has way on municipality	at least one PR repr <sup>7</sup> and on PR block–H	esentative and zero c louse term;	otherwise;	

Municipal and year fixed effects are included.

\* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.

Table 9: Additional Representation and Transfers from National Government to Municipal Governments

Dependent V	Variable: Munici	pal Public Expen	diture minus Loc	al Tax Revenue (	log per capita)	
			Full Sample (F)	Y 1997 - FY 2010		
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3)Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	$0.0121^{***}$ (0.0043)	$0.0118^{***}$ (0.0042)	$0.0122^{***}$ (0.0045)	$0.0117^{***}$ (0.0041)	$\begin{array}{c} 0.0120^{***} \\ (0.0042) \end{array}$	$\begin{array}{c} 0.0109^{***} \\ (0.0040) \end{array}$
R-squared # Obs.	0.9622 $34906$	0.9623 34906	0.9623 $34717$	0.9628 34906	0.9622 34838	0.9632 34649
		Quasi-r	andomized San	aple (FY 1997 - 1	ry 2010)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) (Coeff. / (S.E.))
PR Representative	$0.0230^{***}$ (0.0083)	$0.0226^{***}$ (0.0085)	$0.0208^{**}$ (0.0082)	$0.0229^{***}$ (0.0086)	$0.0256^{***}$ (0.0082)	$0.0232^{***}$ (0.0083)
R-squared # Obs.	0.9747 10208	$0.9748 \\ 10208$	$0.9752 \\ 10156$	0.975 10208	$0.9749 \\ 10184$	0.9758 10132
Control Variables:	(1)	(2)	(3)	(4)	(5)	(9)
Population (log) Taxable Income per capita (log)		Y	Y			Y
Age 0 to $4 /$ Population Age 5 to 19 / Population				YY		Ч
Age $65+$ / Population Size of District Electorate (log)				Υ	Y	Y
<b>PR Representative</b> is a dummy vari Standard errors in parentheses are robi	iable equal to one if t oust to clustering two-	he municipality has way on municipality	at least one PR repr and on PR block–H	esentative and zero o louse term;	otherwise;	

Table 10: Additional Representation and Gap Between Per-Capita Local Public Expenditure and Local Tax Revenue

Municipal and year fixed effects are included.

Lable 11: Additional Kepresent	cation and I	Full Sa	nple	orks, weltar	e and Govern	ment Fayroll
Dependent Variable:	Infrast	ructure	Wel	fare	Ч	ayroll
(expenditure on:)	(1)	(2)	(3)	(4)	(2)	(9)
PR Representative	$0.0251^{*}$	$0.0244^{*}$	-0.0075	-0.0067	0.0014	0.0011
	(0.0140)	(0.0129)	(0.0053)	(0.0051)	(0.0024)	(0.0023)
# Obs.	17368	17149	17368	17149	17368	17149
		mobue R : e	Jume2 bezi			
	c m ð		idinon norr	)		
Dependent Variable:	Infrast	ructure	Wel	fare	Р	ayroll
(expenditure on:)	(1)	(2)	(3)	(4)	(5)	(9)
PR Representative	$0.0770^{**}$	$0.0751^{**}$	$-0.0219^{**}$	$-0.0192^{*}$	0.0057	0.0053
	(0.0336)	(0.0332)	(0.0105)	(0.0101)	(0.0062)	(0.0054)
# Obs.	5956	5888	5956	5888	5956	5888
Controls:	(1)	(2)	(3)	(4)	(5)	(9)
Population (log)		Υ		Υ		Υ
Taxable Income per capita (log)		Y		Y		Υ
Age 0 to $4 /$ Population		Y		Y		Υ
Age 5 to $19 /$ Population		Y		Υ		Υ
Age $65 + /$ Population		Υ		Υ		Υ
Size of District Electorate (log)		Υ		Υ		Υ
<b>PR Representative</b> is a dummy varia	able equal to	one if the muni	cipality has at	least one PR r	epresentatives a	nd zero otherwise;
Each cell contains an estimate of the tr	eatment effec	t from one regi	tession;			

In columns (1) and (2), the dependent variable is log per capita municipal expenditure on public works;

In columns (3) and (4), the dependent variable is log per capita municipal welfare spending; In columns (5) and (6), the dependent variable is log per capita municipal expenditure on government payroll;

Standard errors in parentheses are robust to clustering two-way on municipality and on PR block-House term; \* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.

(Quasi-	Randomized	1 Sample; F + Variable:	<u> - 7991 Y</u> Municini	Public Ext	anditure ner	ranita (loa)
	nehennen	ר אמדומהובי	TAT MITTCI D 41	Turn Turn	iad aminina	capita (108)
	(1)	(2)	(3)	(4)	(5)	(9)
PR Representative	$0.0152^{**}$	$0.0148^{**}$	$0.0130^{*}$	$0.0154^{**}$	$0.0184^{***}$	$0.0161^{***}$
	(0.0068)	(0.0074)	(0.0067)	(0.0070)	(0.0064)	(0.0063)
SMD Rep. in Governing Coalition	$0.0181^{*}$	$0.0192^{*}$	$0.0188^{*}$	$0.0178^{*}$	$0.0147^{*}$	$0.0154^{*}$
	(0.0103)	(0.0112)	(0.0103)	(0.0107)	(0.0084)	(0.0088)
PR Rep. in Governing Coalition	$0.0143^{*}$	$0.0137^{*}$	0.0118	0.0137	0.0116	0.0087
	(0.0081)	(0.0083)	(0.0079)	(0.0085)	(0.0073)	(0.0075)
Population (log)		Υ				Υ
Taxable Income per capita (log)			Υ			Υ
Age 0 to $4 /$ Population				Υ		Υ
Age 5 to $19$ / Population				Υ		Υ
Age $65+$ / Population				Υ		Υ
Size of District Electorate (log)					Υ	Υ
# Obs.	10208	10208	10156	10208	10184	10132
<b>PR Representative</b> is a dummy variable	e equal to one	if the munic	ipality has a	PR represent	tative and zero	otherwise;
Standard errors in narentheses are rohust	to two-way c	histering on r	mnicipality	and on P.B. bl	ock–House tern	

oalition and Per-Capita Local Public Expenditure	$C_{0,000}$ [0, EV 1007 EV 9010]
12: Representatives in Governing C	(Oned Dandamized
Table $12$	

GIIII) rcy. npdr III SIII vay u Numicipal and year fixed effects are rouge to two municipal and year fixed effects are included. \* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.

	Depend	lent Variable	: Municipal	Public Expe	nditure per cap	ita (log)
	(1)	(2)	(3)	(4)	(5)	(9)
PR Representative	$0.0188^{**}$	$0.0231^{***}$	$0.0197^{***}$	$0.0224^{***}$	$0.0210^{***}$	$0.0200^{***}$
PR Representative $\times$ Heterogeneity	(0.0055)	(0.0061) (0.0061)	(0.0042) $(0.0042)$	$0.0168^{**}$ (0.0071)	(0.0053) (0.0053)	0.0105 (0.0066)
Within SMD Heterogeneity in: Population (log)	-0.0098 (0.0133)					
Taxable Income per capita (log)		$0.0165^{**}$				
Age 0 to $4 /$ Population			-0.0030			
Age 5 to 19 $/$ Population			(00000)	-0.0043		
Age $65+$ / Population				(2100.0)	-0.0120	
Municipal Tax Revenue / Municipal Pu	ublic Expenditure				(6000.0)	-0.0095 $(0.0064)$
R-squared	0.9718	0.9719	0.9718	0.9717	0.9718	0.9727
# Obs.	10080	10075	10080	10080	10080	9275

onditure Der Cenite on I cool Dublic Fyn ontotion Dove Dove monoit... Table 13. Within CMD Hater Measures of within SMD heterogeneity are the within SMD standard deviations across municipalities in log population, share of population aged 65+ etc; Measures of within SMD heterogeneity are normalized to have mean zero and standard deviation of one;

Standard errors in parentheses are robust to two-way clustering on municipality and on PR block-House term;

Municipal and year fixed effects are included.

Election
lowing
Fo
the
in
Outcomes
þ
l Performance an
Electoral
Candidates'
4:
Table 1

		SMD Win	ners			
Binary dependent variable indicating the SMD winner's outcome in the next election:	(1) Rur	(2) 1 Again	(3) Be Elected	(4) to an SMD Seat	(5) Be Elect	(6) ted to Any Seat
PR Rep.	-0.0464 $(0.0300)$	-0.0545*(0.0314)	-0.0047 $(0.0458)$	-0.0697*(0.0363)	0.0111 (0.0455)	-0.0161 (0.0406)
Vote Share		-0.0858		1.2872***		$1.5160^{***}$
log(1 + tenure)		$(0.1889) -0.0502^{**}$ (0.0236)		(0.2708) -0.0390 (0.0282)		(0.2818) -0.0538* (0.0290)
Constant	$0.9142^{***}$ (0.0158)	$1.0098^{***}$ (0.0879)	$0.4645^{***}$ (0.0285)	-0.0911 $(0.1335)$	$0.6154^{***}$ (0.0277)	-0.0477 (0.1374)
Fixed Effects	I	Party-Election	I.	Party-Election	I	Party-Election
# Obs.	512	512	512	512	512	512
		SMD Runn	ers-up			
Binary dependent variable indicating the	(1)	(2)	(3)	(4)	(5)	(9)
runner-up's outcome in the next election:	Rur	ı Again	Be Elected	to an SMD Seat	Be Elect	ted to Any Seat
PR Rep.	$0.3441^{***}$	$0.2583^{***}$	$0.1562^{***}$	$0.1804^{***}$	$0.1901^{***}$	$0.1923^{***}$
Vote Share	(0.0374)	(0.0444) $1.3843^{***}$	(0.0420)	(0.0400) 1.3878***	(0.0468)	(0.0463) $1.4894^{***}$
log(1 + tenure)		(0.3885) -0.0690**		(0.3294) -0.0216		(0.3861) -0.0533*
Constant	$0.5237^{***}$ (0.0268)	(0.0287) 0.0917 (0.1401)	$0.2633^{***}$ (0.0245)	(0.0262) - $0.2434^{**}$ (0.1157)	$0.3846^{***}$ (0.0274)	(0.0302) -0.1282 (0.1376)
Fixed Effects		Party-Election	I	Party-Election	I	Party-Election
# Obs.	512	512	512	512	512	512
OLS Regressions in this table examine candidates' el The upper panel presents results for winners of SMD For both nanel candidates are from districts in the o	ectoral participa races. The lowe	ttion and outcomes in er panel presents resu samule described in	a the following g ilts for runners-u Section 3	eneral election. up of SMD races.		

Vote Share is the SMD vote share of the candidate (SMD winners in the upper panel and runner-up in the lower panel). This variable has support between zero and one. In columns (3) & (4), the dependent variable is a dummy variable equal to one if the candidate wins an SMD seat in the next election and zero otherwise. In columns (5) & (6), the dependent variable is a dummy variable equal to one if the candidate wins any seat in the next election and zero otherwise. In columns (1) & (2), the dependent variable is a dummy variable equal to one if the candidate runs in the next election and zero otherwise. In the upper panel, **PR Rep.** is a dummy variable equal to one if the winner's district has a PR representative and zero otherwise. In the lower panel, **PR Rep.** is a dummy variable equal to one if the runner-up is elected to a PR seat and zero otherwise.

log(1 + tenure) is the natural logarithms of one plus the number of terms the candidate has served in the House of Representatives.

Standard errors in the parentheses are robust to clustering on politician. \* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.

Table 15: Municipal Vote Sna	tre Margin ber Dep	bendent Varia	ble: Municipa	<u>nner-up പ്രua</u> l Public Expe	sı-Kandomized nditure per cap	Sample) ita (log)
	(1)	(2)	(3)	(4)	(5)	(9)
PR Representative $(PR)$	$0.0132^{*}$	0.0118	$0.0128^{*}$	0.0096	0.0098	0.0092
	(0.0072)	(0.0074)	(0.0072)	(0.0074)	(0.0079)	(0.0076)
Municipal Margin $(MM)$	0.0221	0.0380	0.0384	0.0344	0.0436	0.0449
	(0.0235)	(0.0261)	(0.0263)	(0.0282)	(0.0280)	(0.0285)
$MM^2$				-0.0597	0.0068	-0.0335
				(0.0574)	(0.0664)	(0.0642)
PR  imes MM	-0.0018	-0.0060	-0.0072	-0.0392	-0.0435	-0.0449
	(0.0271)	(0.0272)	(0.0274)	(0.0319)	(0.0313)	(0.0336)
$PR  imes MM^2$				$0.1775^{**}$	$0.1544^{*}$	$0.1839^{**}$
				(0.0870)	(0.0855)	(0.0901)
District-wide Margin $(DM)$		-0.0524			0.0516	
		(0.0425)			(7700.0)	
$DM^2$					-0.3248	
					(0.2244)	
Vote Share of SMD Winner $(VS_w)$			-0.0949			-0.1025
			(0.0589)			(0.0623)
Vote Share of SMD Runner-up $(VS_r)$			0.0097			0.0045
			(0.0665)			(0.0717)
Fixed Effects	Municipal	Municipal	Municipal	Municipal	Municipal	Municipal
	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year
Weighted	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$
R-squared	0.9744	0.9744	0.9745	0.9745	0.9746	0.9745
# Obs.	8137	8137	8137	8137	8125	8137
<b>PR Representative</b> is a dummy variable eq	lual to one if the	municipality ha	s a PR represent	atives and zero	otherwise;	
Municipal Margin $(MM)$ is the difference	of vote share bet	tween the SMD	winner and the S	SMD runner-up	in the municipality	у;
<b>District-wise Margin</b> $(DM)$ is the difference	ce of vote share l	between the SM	D winner and th	e SMD runner-u	tp in the whole sin	gle-member district;

Vote Share of SMD Winner  $(VS_w)$  is the vote share of the SMD winner in the whole single-member district;

Vote Share of SMD Runner-up  $(VS_r)$  is the vote share of the SMD runner-up in the whole single-member district; Standard errors in parentheses are robust to clustering two-way on municipality and on PR block-House term;

		Dependent Varia	ble: Municipal P	ublic Expenditure	e per capita (log)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
Any PR Representative	0.0084**	0.0084**	0.0078**	0.0082**	0.0086***	0.0071**
Two PR. Representatives	(0.0033)	(0.0033) 0.0063	(0.0034) 0.0101	(0.0033) 0.0029	(0.0032) 0.0065	(0.0030)
	(0.0140)	(0.0140)	(0.0154)	(0.0115)	(0.0139)	(0.0125)
Population (log)		0.0211				0.0912
		(0.0593)				(0.0554)
Taxable Income per capita (log)			$0.1505^{***}$			$0.1943^{***}$
			(0.0513)			(0.0476)
Age 0 to $4 /$ Population				$-1.2547^{**}$		$-1.1052^{**}$
				(0.5191)		(0.5046)
Age 5 to $19 /$ Population				0.0572		0.1384
				(0.3394)		(0.3449)
Age $65 + /$ Population				$1.1507^{***}$		$1.5533^{***}$
				(0.2472)		(0.2734)
Size of District Electorate (log)					-0.0126	-0.0070
					(0.0461)	(0.0464)
Fixed Effects	Municipal	Municipal	Municipal	Municipal	Municipal	Municipal
	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year	Fiscal Year
Weighted	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$
R-squared	0.9596	0.9596	0.9601	0.9600	0.9596	0.9608
# Obs.	34906	34906	34717	34906	34838	34649
Any PR Representative is a dumm	variable equal to or	ie if the municipality	<sup>7</sup> has at least one PI	R representatives and	l zero otherwise;	

Table A.1: Additional Representation and Local Public Expenditure Per Capita (Full Sample)

Two PR Representative is a dummy variable equal to one if the municipality has two PR representatives and zero otherwise; Standard errors in parentheses are robust to clustering two-way on municipality and on PR block-House term;

I	Dependent Varia	able: Municipal	Public Expenditu	ıre per capita (log		
			Full Sample (F <sup>y</sup>	Y 1997 - FY 2010	(	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4)Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	$0.0102^{***}$ (0.0037)	$0.0100^{***}$ (0.0036)	$0.0097^{***}$ (0.0038)	$0.0091^{**}$ (0.0036)	$0.0103^{***}$ (0.0037)	$0.0081^{**}$ (0.0034)
R-squared # Obs.	0.9315 34906	0.9316 34906	0.9325 34717	0.9327 34906	0.9315 34838	0.9345 34649
		Quasi-r	andomized San	aple (FY 1997 - I	FY 2010)	
	(1) Coeff. / (S.E.)	(2) Coeff. / (S.E.)	(3) Coeff. / (S.E.)	(4) Coeff. / (S.E.)	(5) Coeff. / (S.E.)	(6) Coeff. / (S.E.)
PR Representative	$0.0130^{**}$ (0.0064)	$0.0135^{**}$ (0.0066)	$0.0125^{*}$ (0.0066)	$0.0131^{**}$ (0.0064)	$0.0131^{**}$ (0.0064)	$0.0124^{**}$ (0.0063)
R-squared # Obs.	0.9557 10208	0.9559 10208	0.9576 10156	0.9562 10208	0.9561 10184	$0.9584 \\ 10132$
Control Variables:	(1)	(2)	(3)	(4)	(5)	(9)
Population (log) Taxable Income per capita (log)		Υ	Υ			Y
Age 0 to 4 / Population Age 5 to 19 / Population				YY		YY
Age $65+$ / Population Size of District Electorate (log)				Υ	Υ	Y
Population Weighted	Υ	Υ	Υ	Υ	Υ	Υ
PR Representative is a dummy vari. Standard errors in narentheses are roly	iable equal to one if t	he municipality has way on municipality	at least one PR repr , and on PR block_F	esentative and zero o louse term	otherwise;	

f 5 modi Ś ے م 2

Municipal and year fixed effects are included;

Each observation is weighted by the municipal population in the year.

Table A.3: Balanced Full Sample and A Sub-Sample of	f Municipali	ties with Lo	ong History	in the Quas	ii-Randomize	ed Sample
Balanced I	Panel of the	Full Sampl	е			
Dependent Variable:	(1)	(2)	(3)	(4)	(5)	(9)
Public Expenditure (log per capita)	$0.0073^{*}$	$0.0072^{*}$	0.0063	$0.0073^{*}$	0.0078*	0.0060
	(0.0043)	(0.0042)	(0.0044)	(0.0041)	(0.0040)	(0.0037)
Expenditure minus Local Tax Revenue (log per capita)	$0.0122^{**}$	$0.0118^{**}$	$0.0117^{**}$	$0.0123^{**}$	$0.0124^{**}$	$0.0107^{**}$
	(0.0058)	(0.0056)	(0.0059)	(0.0054)	(0.0056)	(0.0051)
# Obs.	19943	19943	19941	19943	19913	19911
A Sub-Sample of Municipalities in the Qusai-Randomi	the Quasi-R ized Sample	andomized for at least	Sample: Half of the	Sample Per	riod	
Dependent Variable:	(1)	(2)	(3)	(4)	(5)	(9)
Public Expenditure (log per capita)	$0.0188^{**}$	$0.0180^{**}$	$0.0180^{**}$	$0.0193^{**}$	$0.0216^{***}$	$0.0209^{***}$
	(0.0081)	(0.0084)	(0.0072)	(0.0084)	(0.0077)	(0.0072)
Expenditure minus Local Tax Revenue (log per capita)	$0.0259^{**}$	$0.0253^{**}$	$0.0253^{***}$	$0.0264^{**}$	$0.0282^{***}$	$0.0281^{***}$
	(0.0101)	(0.0102)	(0.0095)	(0.0107)	(0.0099)	(0.0101)
# Obs.	3706	3706	3701	3706	3705	3700
Controls:	(1)	(2)	(3)	(4)	(5)	(9)
Population (log)		Υ				Υ
Taxable Income per capita (log)			Υ			Υ
Age 0 to $4 /$ Population				Υ		Υ
Age 5 to $19 \ /$ Population				Υ		Υ
Age $65 + /$ Population				Υ		Υ
Size of District Electorate (log)					Υ	Υ
<b>PR Representative</b> is a dummy variable equal to one if the mur	nicipality has a	at least one P	R representat	ive and zero o	therwise;	
Standard errors in parentheses are robust to two-way clustering on	ı municipality	and on PR b	lock–House te	rm;		

Each cell contains an estimate of treatment effect from one regression. Different columns represent specifications with different controls;

Municipal and year fixed effects are included in each regression.

$\frown$
$\mathbf{r}$
2
ıt.
O
$\odot$
Ŀ
ra
2
5
He
Щ
Ч
÷:
≥
14
Se
Ę
$\mathcal{O}$
S
es
q
$\mathbf{x}$
nc
of
Ŕ
Ċ
La.
ji
ਬ
U
GL
ă
යා -
Ĕ
井
÷
ğ
é
Ř
Ē
U
÷
육
Ľ.
Ξ
ģ
ŏ
Ĺ
5
ň
а
Ц
÷
at
nt
ĕ
6.
DI.
ē
Ч
Γſ
nź
<u>io</u>
ij.
ld
V.
4
÷
, ,
4
le
q
ñ

	Fu	ill Sample				
	[]	HS: Municip	al Public Ex	penditure pe	er capita (log	g)
	(1)	(2)	(3)	(4)	(5)	(9)
A PR Representative	$0.0089^{***}$	$0.0086^{***}$	$0.0087^{***}$	0.0087***	$0.0088^{***}$	$0.0089^{***}$
	(0.0032)	(0.0032)	(0.0032)	(0.0032)	(0.0032)	(0.0032)
Vote Share of SMD Winner	0.0115			0.0374	0.0231	0.0475
	(0.0207)			(0.0424)	(0.0348)	(0.0427)
Victory Margin of SMD Winner		0.0013		-0.0193		-0.0559
		(0.0115)		(0.0236)		(0.0448)
Narrowness Ratio of SMD Runner-up			-0.0013		0.0066	-0.0223
			(0.0077)		(0.0131)	(0.0246)
# Obs	34906	34906	34906	34906	34906	34906
	Quasi-Raı	ndomized S	ample			
	[1]	HS: Municip	al Public Ex	penditure pe	er capita (log	
	(1)	(2)	(3)	(4)	(5)	(9)
PR Representative	$0.0180^{***}$	$0.0181^{***}$	$0.0180^{***}$	$0.0185^{***}$	$0.0182^{***}$	$0.0183^{***}$
	(0.0067)	(0.0068)	(0.0068)	(0.0067)	(0.0067)	(0.0067)
Vote Share of SMD Winner	-0.0239			-0.0471	-0.0317	-0.0583
	(0.0466)			(0.0761)	(0.0658)	(0.0802)
Victory Margin of SMD Winner		-0.0043		0.0197		0.0917
		(0.0278)		(0.0462)		(0.1079)
Narrowness Ratio of SMD Runner-up			0.0048		-0.0047	0.0446
			(0.0184)		(0.0261)	(0.0601)
# Obs	10208	10208	10208	10208	10208	10208
<b>PR Representative</b> is a dummy variable equ	al to one if th	e municipality	has at least or	ne PR represen	tative and zero	otherwise:

Standard errors in parentheses are robust to two-way clustering on municipality and on PR block-House term; Municipal and year fixed effects are included. \* p < 0.10; \*\* p < 0.05; \*\*\* p < 0.01.